\title{\bf Connectivity for random graphs\\ from a weighted bridge-addable class}
\documentclass[12pt]{article}
\usepackage{e-jc}
\specs{P53}{19(4)}{2012}
\sloppy

\usepackage{amsthm,hyperref}

\newcommand{\doi}[1]{\href{http://dx.doi.org/#1}{\texttt{doi:#1}}}
\newcommand{\arxiv}[1]{\href{http://arxiv.org/abs/#1}{\texttt{arXiv:#1}}}
 
  \usepackage{amsmath}
  \usepackage{amssymb}
  \usepackage{latexsym}
  \usepackage{graphicx}
\allowdisplaybreaks

%  \newenvironment{proof}{\noindent{\bf Proof} \hspace{.1in}}{\hspace*{\fill}$\Box$}
  \newenvironment{proofof}[1]{%
  \noindent {\bf Proof of #1}}%
  {\hspace*{\fill}$\Box$}
  \def\noproof{\hspace*{\fill}$\Box$}

  \newtheorem{theorem}{Theorem}[section]
  \newtheorem{lemma} [theorem] {Lemma}%[section]
  \newtheorem{corollary} [theorem] {Corollary}%[section]
  \newtheorem{prop} [theorem] {Proposition}%[section]
  \newtheorem{definition} [theorem] {Definition}%[section]
  \newtheorem{remark} [theorem] {Remark}%[section]
  \newtheorem{conjecture} [theorem] {Conjecture}%[section]
  \newtheorem{obs}[theorem]{Observation}
   
\def\Prob#1{\mbox{{\rm Prob}}\,[#1]}
\def\Exp{{\mathbb E}\,}
\def\Pr{\mbox{{\rm Pr}}\,}
\let\eps=\epsilon
\def\enddiscard{}
\long\def\discard#1\enddiscard{}
\newcommand{\?}{?\marginpar[\hfill ?]{?}}

  \def\Po{\mbox{\rm Po}}
  \newcommand{\bc}{{\rm BC}}
  \newcommand{\CP}{\rm CP}
  %\newcommand{\eps}{\varepsilon}
  \newcommand{\E}{\mathbb E}
  \newcommand{\pr}{\mathbb P}
  \newcommand{\giant}{\rm giant}
  \newcommand{\Bigc}{\rm Big}
  \newcommand{\bigc}{\rm big}
  %\newcommand{\Miss}{\rm Miss}
  %\newcommand{\miss}{\rm miss}
  \newcommand{\eg}{\rm eg}
  \newcommand{\inu}{\in_{u}}
  \newcommand{\inl}{\in_{\lambda}}
  \newcommand{\intau}{\in_{\tau}}
  \newcommand{\whp}{whp }
  
  \newcommand{\mass}[1]{\mbox{mass}(#1)}


  \newcommand{\m}[1]{\marginpar{\tiny{#1}}}
  \newcommand{\gc}{\gamma_{\ell}}
  \newcommand{\gu}{\gamma_{u}}
  \newcommand{\gs}{{\mathcal G}^{S}}
  \newcommand{\as}{{\mathcal A}^{S}}
  \newcommand{\bs}{{\mathcal B}^{S}}
  \newcommand{\cs}{{\mathcal C}^{S}}
  \newcommand{\mgs}{|{\mathcal G}^{S}_n|}
  \newcommand{\cp}{{\mathcal P}}
  \newcommand{\cq}{{\mathcal Q}}
  
  \newcommand{\ca}{{\mathcal A}}
  \newcommand{\cb}{{\mathcal B}}
  \newcommand{\cc}{{\mathcal C}}
  \newcommand{\cd}{{\mathcal D}}
  \newcommand{\ce}{{\mathcal E}}
  \newcommand{\cf}{{\mathcal F}}
  \newcommand{\cg}{{\mathcal G}}
  \newcommand{\ct}{{\mathcal T}}
  \newcommand{\cu}{{\mathcal U}}
  \newcommand{\cv}{{\mathcal V}}
  \newcommand{\rU}{\rm U}
  
  \newcommand{\RF}{R^{\cf}}
  \newcommand{\RT}{R^{\ct}}
  %\newcommand{\RG}{R^{G}}
  \newcommand{\RG}{R^{G_0}}

  \newcommand{\ugs}{{\mathcal U}{\mathcal G}^{S}}
  \newcommand{\umgs}{|{\mathcal U}{\mathcal G}^{S}_n|}
  \newcommand{\up}{{\mathcal U}{\mathcal P}}
  \newcommand{\rhoP}{\rho_{\cal P}}
  \newcommand{\UP}{U_{\cal P}}

  \newcommand{\add}{{\mbox{add}}}
  \newcommand{\aut}{\mbox{\small aut}}
  \newcommand{\nono}{\Nats}
  \newcommand{\core}{2\!-\!{\rm core}}
   \newcommand{\ex}{{\rm Ex}}

  \newcommand{\Frag}{\mbox{{\rm Frag}}}
  \newcommand{\frag}{{\mbox{\rm frag}}}
  \newcommand{\wfrag}{{\mbox{\rm wfrag}}}

  \newcommand{\Bridge}{{\rm Bridge}}
  \newcommand{\bridge}{{\rm bridge}}
  \newcommand{\Cross}{{\rm Cross}}
  \newcommand{\cross}{{\rm cross}}
  \newcommand{\apex}{{\rm Apex}}
  \newcommand{\ttau}{\tilde{\tau}}

  \newcommand{\Go}{G^{o}}
  \newcommand{\cao}{\ca^{o}}

                                                        
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\begin{document}
\begin{thebibliography}{18}

%\m{add ref for rooted forests}

\bibitem{amr2011} L. Addario-Berry, C. McDiarmid and B. Reed.
  \newblock
  Connectivity for bridge-addable monotone graph classes.
  \newblock
  {\em Combinatorics, Probability and Computing}, first view online, July 2012.
  %2011, arXiv:1110.0009.

 %\bibitem{amr08} \m{update}
 % L. Addario-Berry, C. McDiarmid and B. Reed,
 % Connectivity of Addable Monotone Graph Classes, 2008.
  %submitted 12-05-08 to CPC
    
\bibitem{bbg07}
  P. Balister, B. Bollob{\'a}s and S. Gerke.
  \newblock
  {Connectivity of addable graph classes}. 
  \newblock
  {\em J. Combin. Th. B} {\bf 98} (2008) 577 -- 584.
  
\bibitem{bbg10}
  P. Balister, B. Bollob{\'a}s and S. Gerke.
  \newblock
  Connectivity of random addable graphs.
  \newblock
  {\em Proc. ICDM 2008} No 13 (2010) 127 -- 134.
    
%\bibitem{dvw96}
%  A.~Denise, M.~Vasconcellos and D.~Welsh,
%  The random planar graph,
%  {\em Congressus Numerantium }{\bf 113} (1996) 61--79.
  
\bibitem{drmota2009}
  M. Drmota.
  \newblock {\em Random Trees}.
  \newblock Springer, 2009. 
 
%\bibitem{fs09}
%  P. Flajolet and R. Sedgewick,
%  {\em Analytic Combinatorics}, CUP, 2009.
     
% \bibitem{gmsw05}
%   S. Gerke, C. McDiarmid, A. Steger and A. Weissl,
%   Random planar graphs with n nodes and a fixed number of edges,
%  {\em  Proceedings of the ACM-SIAM Symposium on Discrete Algorithms (SODA)}
%  2005, 999 -- 1007.

%\bibitem{gmsw07}
%  S. Gerke, C. McDiarmid, A. Steger and A. Weissl,
%  Random planar graphs with given average degree,
%  in {\em Combinatorics, Complexity and Chance,
%  a tribute to Dominic Welsh} (G. Grimmett and C. McDiarmid eds)
%  Oxford University Press, 2007, 83 -- 102.
  
%\bibitem{gn09a}
%  O. Gim\'enez and M. Noy,
%  Asymptotic enumeration and limit laws of planar graphs.
%  %arXiv: math.CO/0501269, 2005  arXiv: math.CO/0501269.
%  {\em J. Amer. Math. Soc.} {\bf 22} (2009) 309--329.

\bibitem{gn09b}
  O. Gim\'enez and M. Noy.
  \newblock
  Counting planar graphs and related families of graphs.
  \newblock
  In {\em Surveys in Combinatorics 2009}, 169 -- 329, Cambridge University Press, Cambridge, 2009.

\bibitem{grimmett06}
  G. Grimmett.
  \newblock {\em The Random-Cluster Model}.
  \newblock Springer, 2006.

%\bibitem{jlr00}
%  S. Janson, T. {\L}uczak and A. Ruci{\'n}ski,
%  {\em Random Graphs}, Wiley Interscience, 2000.

 
%\bibitem{kmcd11}
%  M. Kang and C. McDiarmid,
%  Random unlabelled graphs containing few disjoint cycles,
%  {\em Random Structures and Algorithms} {\bf 38} (2011) 174 -- 204.
  
\bibitem{kp2011}
  M. Kang and K. Panagiotou.
  \newblock  On the connectivity of random graphs from addable classes, 2011.
  %\m{arxiv?}

%\bibitem{km08b}
%  Kurauskas, V. and McDiarmid, C.  (2008)
%  Random graphs with few disjoint cycles.
%  {\em Combinatorics, Probability and Computing}, published online 9 June 2011.
  
%\bibitem{km11}
%  Kurauskas, V. and McDiarmid, C.  (2011)
%  Random graphs containing few disjoint excluded minors. 

%\m{Lovasz, problem 4.3}
%\bibitem{lovaszbook}
%  L. Lov\'asz,
%  {\em Combinatorial Problems and Exercises},
%  North-Holland, 2nd edition, 1993.
  
  
%\bibitem{luczak98}
%  T. {\L}uczak,
%  Random trees and random graphs,
%  {\em Random Structures and Algorithms} {\bf 13} (1998) 485 -- 500. \m{wanted?}

%\bibitem{lp92}
%  T. {\L}uczak and B. Pittel.
%  Components of random forests,
%    {\em Combinatorics, Probability and Computing} {\bf 1} (1992) 35 -- 52. \m{wanted?}

\bibitem{cmcd08}
  C.~McDiarmid.
  \newblock Random graphs on surfaces.
  \newblock {\em J. Combinatorial Theory B} {\bf 98} (2008) 778 -- 797.

\bibitem{cmcd09}
  C.~McDiarmid.
  \newblock Random graphs from a minor-closed class.
  \newblock {\em Combinatorics, Probability and Computing} {\bf 18} (2009) 583 -- 599.
  
\bibitem{cmcd-rwg2012}
  C.~McDiarmid.
  \newblock Random graphs from a weighted minor-closed class.
  In preparation, see \arxiv{1210.2701}.  %, on 10-10-2012.
  %\arxiv{1210.2701}
  
%\bibitem{cmcd-unlab2012}
%  C.~McDiarmid, 
%  Minor-closed unlabelled graph classes,
%  in preparation.
  
%\bibitem{cb08}
%   C.~McDiarmid and B. Reed,
%   On the maximum degree of a random planar graph,
%   {\em Combinatorics, Probability and Computing} {\bf 17} (2008) 591-- 601.

\bibitem{msw05}
  C.~McDiarmid, A.~Steger and D.~Welsh.
  \newblock Random planar graphs.
  \newblock {\em J. Combinatorial Theory B} {\bf 93} (2005) 187 -- 206.

\bibitem{msw06}
  C.~McDiarmid, A.~Steger and D.~Welsh.
   \newblock Random graphs from planar and other addable classes.
   \newblock In {\em Topics in Discrete Mathematics} (M. Klazar, J. Kratochvil,
  M. Loebl, J. Matousek, R. Thomas, P. Valtr, Eds.),
  Algorithms and Combinatorics 26, Springer, 2006, 231 -- 246.

\bibitem{moon67}
  J.W.~Moon.
   \newblock {Enumerating labelled trees}.
   \newblock In {\em Graph Theory and Theoretical Physics} (F. Harary, Ed.),
 Academic Press, San Diego, 1967, 261 -- 272. 

\bibitem{moon70}
  J.W.~Moon.
   \newblock {\em Counting labelled trees}.
   \newblock {Canadian Mathematical Monographs} {\bf 1}, 1970.

\bibitem{sn2012}
  S.~Norine.   \newblock Private communication, 2012.
  
%\bibitem{ps10}
%  K. Panagiotou and A. Steger,
%  Maximal biconnected subgraphs of random planar graphs,
%  {\em ACM Transactions on Algorithms} {\bf 6} (2010) art. no. 31.
%  \m{2010?}

\bibitem{pitman99}
  J. Pitman.
   \newblock Coalescent random forests.
   \newblock {\em J. Combinatorial Theory A} {\bf 85} (1999) 165 -- 193.
    
\bibitem{renyi59}
   A. R\'enyi. 
   \newblock  Some remarks on the theory of trees.
    \newblock {\em Publications of the Mathematical Institute of the
   Hungarian Academy of Sciences} {\bf 4} (1959) 73 -- 85.

\bibitem{renyi1970}
  A. R\'enyi.
   \newblock On the enumeration of trees.
   \newblock {\em Combinatorial Structures and their Applications},
  R. Guy, H. Hanani, N. Sauer and J. Schonheim (Eds), Gordon and Breach, New York, 1970, 355 -- 360.
 
\end{thebibliography}


\author{Colin McDiarmid \\ 
\small Department of Statistics \\[-0.8ex] 
\small Oxford University, UK}
\date{\dateline{Aug 1, 2012}{Dec 3, 2012}{Dec 31, 2012}\\
\small Mathematics Subject Classifications: 05C80, 05C40}


%\author{Colin McDiarmid}
\maketitle

\begin{abstract}
  There has been much recent interest in random graphs sampled uniformly from the
  $n$-vertex graphs in a suitable structured class, such as the class of all planar graphs.
  Here we consider a general \emph{bridge-addable} class $\ca$ of graphs --
  if a graph is in $\ca$ and $u$ and $v$ are vertices in different components
  then the graph obtained by adding an edge (bridge) between $u$ and $v$ must also be in $\ca$. 
  Various bounds are known concerning the probability of a random graph from such a
  class being connected or having many components, sometimes under the additional assumption that bridges
  can be deleted as well as added.
  Here we improve or amplify or generalise these bounds (though we do not resolve the main conjecture).
  For example, we see that the expected number of vertices left when we remove a largest component is less than $2$.
  The generalisation is to consider `weighted' random graphs,
  sampled from a suitable more general distribution,
  where the focus is on the bridges.
  %
%  Here we consider a more general model: we consider weighted random graphs from a
%  bridge-addable class of graphs, where we give weights to edges and numbers of components.
%  We find that earlier results on the probability of connectedness extend naturally to the general weighted framework,

 \bigskip\noindent \textbf{Keywords:}
  random graph, connectivity, components, bridge-addable class
\end{abstract}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\section{Introduction}
\label{sec.intro}

  A \emph{bridge} in a graph is an edge $e$ such that the graph $G \setminus e$ obtained by deleting $e$ has
  one more component. 
  A class $\ca$ of graphs is {\em bridge-addable} if for all graphs $G$ in $\ca$ and all vertices
  $u$ and $v$ in distinct connected components of $G$, the graph $G +uv$ obtained by adding an
  edge between $u$ and $v$ is also in $\ca$.
  The concept of being bridge-addable (or {`weakly addable'}) was introduced in 
  McDiarmid, Steger and Welsh~\cite{msw05} in the course of studying random planar graphs.
  %\m{ignore survey unless..}
  (For an overview on random planar graphs see the survey paper~\cite{gn09b} of Gim\'enez and Noy.)
  Examples of bridge-addable classes of graphs include forests, series-parallel graphs,
  planar graphs, and indeed graphs embeddable on any given surface.
  
  In the rest of this section, we first describe what is known concerning connectedness and components for
  random graphs sampled uniformly from a bridge-addable class;
  then describe the new results here for such random graphs; and finally briefly discuss random rooted graphs.
  Random graphs from a weighted class are introduced in Section~\ref{sec.rwg}; and new general results are presented,
  which extend the results on uniform random graphs.
  After that come the proofs, first for non-asymptotic results then for asymptotic results and finally for the rooted case.  
  \medskip
  
  \noindent
  {\em Background results for uniform random graphs}
  
  If $\ca$ is finite and non-empty we write $R \inu \ca$ to mean that $R$ is a random graph sampled
  {\bf u}niformly from $\ca$.  (We consider graphs to be labelled.)
  %
  The basic result on connectivity for a bridge-addable set of graphs is
  Theorem 2.2 of~\cite{msw05}:
  if $\ca$ is a finite bridge-addable set of graphs and $R \in_u \ca$ then
\begin{equation} \label{eqn.b-a-conn}
  \pr(R \mbox{ is connected}) \geq e^{-1} .
\end{equation}
  Indeed, a stronger result is given in~\cite{msw05}, concerning the number $\kappa(R)$ of components of $R$;
  namely that $\kappa(R)$ is stochastically at most $1+\Po(1)$
  where $\Po(\lambda)$ denotes a Poisson-distributed random variable with mean $\lambda$,
  that is
\begin{equation} \label{eqn.b-a-comps}
  \kappa(R) \leq_s 1+ \Po(1).
\end{equation}
  (Recall that $X \leq_s Y$ means that $\pr(X \leq t) \geq \pr(Y \leq t)$ for each $t$.)
  Note that from~(\ref{eqn.b-a-comps}) we have
\[   \pr(R \mbox{ is connected}) = \pr(\kappa(R) \leq 1) \geq \pr(\Po(1) \leq 0) = e^{-1}\]
  and we obtain~(\ref{eqn.b-a-conn}).
  Also from~(\ref{eqn.b-a-comps}) we have $\E[\kappa(R)] \leq 2$ (see~(\ref{eqn.exp}) below).

  For any set $\ca$ of graphs, we let $\ca_n$ denote the set of graphs in $\ca$ on vertex set $[n]:=\{1,\ldots,n\}$;
  and we write $R_n \inu \ca$ to mean that $R_n$ is uniformly distributed over $\ca_n$.
  (It is convenient to have the subscript $n$ on $R$ rather than on $\ca$.)
  We always assume that $\ca_n$ is non-empty at least for large $n$.
  
  The class $\cf$ of forests is of course bridge-addable. For $R_n \in_u \cf$ a result of R\'enyi~\cite{renyi59} shows that
  $\pr(R_n \mbox{ is connected}) \to e^{-\frac12}$ as $n \to \infty$, and indeed
  %\m{in Renyi's paper?}
  $\kappa(R_n)$ converges in distribution to $1+ \Po(\frac12)$.
  For background on random trees and forests see the books~\cite{drmota2009,moon70}.
  It was noted in~\cite{msw06} that plausibly forests form the `least connected' bridge-addable set of graphs,
  and in particular it should be possible to improve the bound in~(\ref{eqn.b-a-conn}) asymptotically.
  %
\begin{conjecture} \cite{msw06} \label{conj.b-add}
  If $\ca$ is bridge-addable and $R_n \in_u \ca$ then
\begin{equation} \label{eqn.b-a_conj}
  \liminf_{n \to \infty} \pr(R_n \mbox{ is connected}) \geq e^{-\frac12}.
\end{equation}
\end{conjecture}  


  Balister, Bollob{\'a}s and Gerke~\cite{bbg07,bbg10} showed that inequality~(\ref{eqn.b-a_conj}) holds
  if we replace $e^{-\frac12} \approx 0.6065$ by the weaker bound $e^{-0.7983} \approx 0.4542$.
  This result has recently been improved by Norine~\cite{sn2012}. %\m{say more?}
  Recently Addario-Berry, McDiarmid and Reed~\cite{amr2011}, and Kang and Panagiotou~\cite{kp2011},
  separately showed that~(\ref{eqn.b-a_conj}) holds with the desired lower
  bound $e^{-\frac12}$, if we suitably strengthen the condition on $\ca$. %to being bridge-alterable:
  Call a set $\ca$ of graphs {\em bridge-alterable} if it is bridge-addable and also closed under
  deleting bridges. Thus $\ca$ is bridge-alterable exactly when it satisfies the condition that,
  for each graph $G$ and bridge $e$ in $G$, the graph $G$ is in $\ca$ if and only if $G \setminus e$ is in $\ca$.
  Observe that each of the bridge-addable classes of graphs mentioned above is in fact bridge-alterable.
  %where $G \setminus e$ denotes the graph obtained by deleting $e$.
  If $\ca$ is a bridge-alterable set of graphs and $R_n \in_u \ca$ then~\cite{amr2011,kp2011} 
\begin{equation} \label{eqn.b-alt}
  \liminf_{n \to \infty} \pr(R_n \mbox{ is connected}) \geq e^{-\frac12}.
\end{equation}
  Since the class $\cf$ of forests is bridge-alterable, this result is best-possible for a bridge-alterable
  set of graphs.  The full version of Conjecture~\ref{conj.b-add} (for a bridge-addable set) is still open.

  Next let us consider the `fragment' of a graph $G$: % not in its largest component:
  we let $\frag(G)$ be the number of vertices remaining when we remove a largest component.
  %(where we break ties by choosing the lex-first largest component);
  %and let $\frag(G)$ be the number of vertices in this subgraph.
  %of $G$ less the maximum number of vertices in  a component.
  For the class $\cf$ of forests, if $R_n \in_u \cf$ then
  %(see for example %~\cite{cmcd-rwg2012} or see Theorem~\ref{prop.asympt-forests} below)
\begin{equation} \label{eqn.forests-frag}  
  \E[\frag(R_n)] \to 1 \;\; \mbox{ as } n \to \infty.
\end{equation}
  It was shown in~\cite{cmcd09} that,
  if $\ca$ is a bridge-addable class of graphs which satisfies the further condition
  that it is closed under forming minors (and so $\ca$ is bridge-alterable), 
  then there is a constant $c=c(\ca)$ such that, for $R_n \in_u \ca$ 
\begin{equation} \label{eqn.b-alt-frag1}
  \E[\frag(R_n)] \leq c \;\; \mbox{ for all } n.
\end{equation}
  \smallskip
  
  \noindent
  {\em New results for uniform random graphs}
  
  In the present paper we much improve inequality~(\ref{eqn.b-alt-frag1}) and extend all the above results to more
  general distributions (similar to distributions considered in~\cite{cmcd-rwg2012}),
  though we continue to consider uniform random graphs in this section.
  (All the results presented here are special cases of results discussed in the following section.)
  
  In particular we see that, if $\ca$ is \emph{any} bridge-addable class of graphs
  (with no further conditions) and $R_n \in_u \ca$, then
\begin{equation} \label{eqn.b-add-frag}
  \E[\frag(R_n)] < 2 \;\; \mbox{ for all } n;
  \end{equation}
  and if $\ca$ is bridge-alterable then 
\begin{equation} \label{eqn.b-alt-frag2}
  \limsup_{n \to \infty} \E[\frag(R_n)] \leq 1.
\end{equation}
  Observe from the limiting result~(\ref{eqn.forests-frag}) that this last bound is optimal for a bridge-alterable
  set of graphs, but perhaps it holds for any bridge-addable set of graphs-- see Section~\ref{sec.concl}.
%\begin{conjecture} \label{conj.b-alt-frag2}
%  If $\ca$ is bridge-addable and $R_n \in_u \ca$ then
%\[ \limsup_{n \to \infty} \E[\frag(R_n)] \leq 1. \]
%\end{conjecture}

  We also strengthen the inequality~(\ref{eqn.b-alt}) in much the same way that the inequality~(\ref{eqn.b-a-comps})
  strengthens~(\ref{eqn.b-a-conn}).
  Given non-negative integer-valued random variables $X_1,X_2,\ldots$ and $Y$, we say that
  $X_n$ is \emph{stochastically at most} $Y$ \emph{asymptotically}, and write %$X_n \leq_s Y$
  $X_n \preceq_s Y$ %$X_n \leq_{sa} Y$
  \emph{asymptotically}, if for each fixed $t \geq 0$,
\[  \limsup_{n \to \infty} \pr(X_n \geq t) \leq \pr(Y \geq t). \]
(Note that, as well as using the word `asymptotically', we write $\preceq_s$ rather than $\leq_s$ here.)
 Our strengthening of~(\ref{eqn.b-alt}) is that, if $\ca$ is bridge-alterable and $R_n \inu \ca$, then 
\begin{equation} \label{eqn.b-alt-comps1}
  \kappa(R_n) \preceq_s 1+ \Po(\frac{1}{2}) \;\; \mbox{ asymptotically }.
\end{equation}
  %; that is, for each fixed integer $t \geq 0$
%\begin{equation} \label{eqn.unif-asympt-comps}
%  \limsup_{n \to \infty} \pr(\kappa(R_n) \geq t+1) \leq \pr(\Po(\frac{1}{2}) \geq t).
%\end{equation} 
  %\smallskip
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%  
  \medskip
  
\noindent
{\em Random rooted graphs}
  
  It may be enlightening to consider rooted graphs, where we find contrasting behaviour.
  We say that a graph is {\em rooted} if each component has a specified root vertex.
  Such graphs are often met in combinatorial enumeration: for example,
  the planar graphs are obtained by starting with a planar graph $G$ which has minimum degree at least two, and
  substituting a rooted tree for each vertex of $G$.
  
  We will use the notation $\Go$ for a rooted graph;
  and given a class $\ca$ of graphs we write $\cao$ for the corresponding class of rooted graphs.
  Thus a connected graph in $\ca_n$ yields $n$ rooted graphs in the corresponding set $\cao_n$;
  a graph in $\ca_n$ which has two components, with respectively $a$ and $n-a$ vertices, yields $a(n-a)$ rooted graphs
  in $\cao_n$; and so on.
  We use the notations $R^{o} \inu \cao$ and $R_n^{o} \inu \cao$ as before, to indicate that
  $R^{o}$ is sampled uniformly from $\cao$ (assumed finite) and $R_n^{o}$ is uniformly sampled from $\cao_n$.
  
  Now let $\ca$ be a finite bridge-addable set of graphs, and let $R^{o} \inu \cao$.
  Since a graph with several non-singleton components generates many rooted graphs,
  it is not immediately clear to what extent the earlier results on connectedness and components will survive.
  We will see that the analogues of~(\ref{eqn.b-a-conn}) and~(\ref{eqn.b-a-comps}) both hold:
\begin{equation} \label{eqn.rooted-conn}
  \pr(R^{o} \mbox{ is connected}) \geq e^{-1}
\end{equation}
  and indeed
\begin{equation} \label{eqn.rooted-comps}
  \kappa(R^{o}) \leq_s 1+ \Po(1).
\end{equation}
  %
  Now consider the class $\cf$ of forests, and let $R_n^{o} \inu \cf^{o}$.
  Then as $n \to \infty$
\[ \pr(R_n^{o} \mbox{ is connected}) = (\frac{n}{n+1})^{n-1} \to e^{-1}\]
  and indeed $\kappa(R_n^{o})$ converges in distribution to $1+\Po(1)$.
  % (see Theorem~\ref{prop.rooted-forests} below).
  Thus~(\ref{eqn.rooted-conn}) and~(\ref{eqn.rooted-comps}) are best possible, in contrast to the unrooted case.
  Further, %in the case of forests, 
  $\E[\frag(R_n^{o})] \to \infty$ as $n \to \infty$,
  so there is no analogue for~(\ref{eqn.b-add-frag}) or~(\ref{eqn.b-alt-frag2}) for rooted graphs.
  %\m{$\frag(R_n^{o})$ is tight for rooted forests, but what about the b-add case?}
  \bigskip
  
  
  In all these results the crucial feature is the behaviour of the bridges.
  We shall bring this out by singling out bridges in the more general distributions we next
  introduce for our random graphs.
  
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


  
\section{Random weighted graphs} \label{sec.rwg}
 
  Given a graph $G$ with vertex set $V$, let $e(G)$ denote the number of edges,
  let $e_0(G)$ denote the number of bridges (edges in 0 cycles) and let
  $\tilde{G}$ denote the graph on $V$ obtained from $G$ by removing all bridges.
  Thus $\kappa(\tilde{G})= \kappa(G)+e_0(G)$.
  
  %Let us call a graph function $f(G)$ \emph{multiplicative} if $f(G)$ equals the
  %product of $f(H)$ over its connected components.
  %(For example let $f(G)= \alpha^{v(G)} \beta^{e(G)} \gamma^{\kappa(G)}$
  %for given $\alpha, \beta, \gamma >0$.) 
  %
  Let $\lambda >0$ and $\nu >0$, and let $f(G) \geq 0$ for each bridge-free graph $G$.
  We call $(\lambda,\nu,f)$ a \emph{weighting} and define the \emph{weight} $\tau(G)$ of $G$
  by setting
\begin{equation} \label{eqn.probdef}
  \tau(G) = f(\tilde{G}) \, \lambda^{e_0(G)} \nu^{\kappa(G)}.
\end{equation}
  Given a set $\ca$ of graphs, let $\tau(\ca)$ denote $\sum_{G \in \ca} \tau(G)$.
  When $0<\tau(\ca) < \infty$, we let $R \in_{\tau} \ca$ mean that $R$ is a random graph sampled from $\ca$ with
  $\pr(R=G) = \tau(G)/ \tau(\ca)$ for each graph $G \in \ca$.
  Also we write $R_n \in_{\tau} \ca$ to mean $R_n \in_{\tau} \ca_n$, as in the uniform case.
  %  Similarly $R_n \in_{\tau} \ca$ means that $R_n$ is a random graph sampled from $\ca_n$ with
  %$\pr(R=G) = \tau(G)/ \tau(\ca_n)$ for each graph $G \in \ca_n$ (and we assume that $\tau(\ca_n)>0$).
  
  In the special case when $\lambda=\nu=1$ and $f(G) \equiv 1$, clearly $R \in_{\tau} \ca$ and
  $R_n \in_{\tau} \ca$ mean the same as $R \inu \ca$ and $R_n \inu \ca$ respectively.
  When $f(G)\equiv \lambda^{e(G)}$ we have $\tau(G) = \lambda^{e(G)} \nu^{\kappa(G)}$,
  and we do not single out bridges.  This case corresponds to the Erd\H{o}s-R\'enyi random graph $G_{n,p}$
  when $\nu =1$ (with $\lambda=p/(1-p)$),
  and in general it corresponds to the random-cluster model (see for example~\cite{grimmett06}).
  
 % Recall that the classical Erd\H{o}s-R\'enyi (or binomial) random graph $G_{n,p}$ has vertex set $[n]$,
 % and the ${n \choose 2}$ possible edges are included independently with probability $p$,
 % where $0<p<1$. Assuming that $\ca_n$ is non-empty, for each $H \in \ca_n$ we have
 % \[
 %   \pr(G_{n,p}=H | G_{n,p} \in \ca ) =
 %   \frac{p^{e(H)}(1-p)^{{n \choose 2}-e(H)}}
 %   {\sum_{G \in \ca_n}p^{e(G)}(1-p)^{{n \choose 2}-e(G)}}
 % =
 %  \frac{\lambda^{e(H)}}
 %   {\sum_{G \in \ca_n}\lambda^{e(G)}}
%\]
 % where $\lambda=p/(1-p)$.
  %Here $e(G)$ denotes the number of edges in $G$: we will use
  %$v(G)$ similarly to denote the number of vertices in $G$.
  %
 % Now consider the more general random-cluster model (see for example~\cite{grimmett06}),
 % where we are also given $\nu>0$ (we use $\nu$ rather than $q$), and
 % the random graph $R_n$ takes as values the graphs $H$ on $[n]$, with
%\[
 % \pr(R_n=H) \propto p^{e(H)}(1-p)^{{n \choose 2}-e(H)} \nu^{\kappa(H)}.
%\]
  %Here $\kappa(H)$ denotes the number of components of $H$.
  %Assuming as before that $\ca_n$ is non-empty,
%  Then for each $H \in \ca_n$
%  \[
%    \pr(R_n=H | R_n \in \ca )
    %= \frac{p^{e(H)}(1-p)^{{n \choose 2}-e(H)} q^{\kappa(H)} }
    %{\sum_{G \in \ca_n}p^{e(G)}(1-p)^{{n \choose 2}-e(G)} q^{\kappa(G)}}
%  =
%   \frac{\lambda^{e(H)}\nu^{\kappa(H)}}
%    {\sum_{G \in \ca_n}\lambda^{e(G)} \nu^{\kappa(G)}} = \frac{\tau(H)}{\tau(\ca_n)}
%\]
%  where $\tau(H)= \lambda^{e(H)} \nu^{\kappa(H)}$, as we met above.
%  \medskip
  
  % Let $f(G)>0$ for each bridgeless graph $G$. 
  %Let us call $\tau=(\lambda,\nu, f)$ a {\em weighting}, and
  %set $\tau(G) = f(b(G)) \cdot \lambda^{e_0(G)} \nu^{\kappa(G)}$.
  %Given a finite set $\ca$ of graphs we let $\tau(\ca) = \sum_{G \in \ca} \tau(G)$.
  %The notation $R \in_{\tau} \ca$ means that $R$ is a random graph in $\ca$, with
  %$\pr(R = G) = \tau(G)/\tau(\ca)$ for each $G \in \ca$.
  %We define a general distribution on a finite set $\ca$ of graphs by setting
  
  Suppose now that we are given a set $\ca$ of graphs and a weighting $(\lambda,\nu,f)$,
  and that $0<\tau(\ca)<\infty$ or $\tau(\ca_n) >0$ as appropriate. 
  We generalise and sometimes amplify all the results presented in the last section.
  %~(\ref{eqn.b-a-conn}), (\ref{eqn.b-a-comps}), (\ref{eqn.b-alt}) and~(\ref{eqn.b-alt-frag1}).
  For the asymptotic results %we will need two further assumptions.
  we need to assume that $\ca$ is bridge-alterable rather than just bridge-addable.
 % and we will assume that the function $f(G)$ is \emph{multiplicative},  
 % %Let us call a graph function $f(G)$ \emph{multiplicative} if 
 % that is, $f(G)$ equals the product of $f(H)$ over its connected components.
 % (For example let $f(G)= \alpha^{v(G)} \beta^{e(G)} \gamma^{\kappa(G)}$
 % for given $\alpha, \beta, \gamma >0$, where $v(G)$ denotes the number of vertices in $G$.) 
 % \smallskip
  
  We first state two non-asymptotic results; then present some results on random forests,
  and consider asymptotic results; and finally we consider random rooted graphs.
  The first result generalises the inequalities~(\ref{eqn.b-a-conn}) and~(\ref{eqn.b-a-comps}), and 
  is used several times in~\cite{cmcd-rwg2012}; and the second result generalises inequality~(\ref{eqn.b-add-frag}).
  %(which improves on the earlier inequality given here as~(\ref{eqn.b-alt-frag1})).

\begin{theorem} \label{prop.tauconn}
  %Let the finite non-empty weighted set $\ca,\tau$ of graphs be bridge-addable,
  %Let $\lambda>0$, $\nu>0$ and $\tau=(\lambda,\nu)$.
  If $\ca$ is finite and bridge-addable and $R \in_{\tau} \ca$, then
  %that is, let the random graph $R$ take values in $\ca$,
  %with $\pr(R=H)= \lambda^{e(H)} \nu^{\kappa(H)}/\tau(\ca)$ for each $H \in \ca$.
\[ \kappa(R) \leq_s 1+ \Po(\nu/\lambda); \]
  and in particular $\; \pr(R \mbox{ is connected}) \geq e^{-\nu/\lambda}$, and
  $\E[\kappa(R)] \leq 1+ \nu/\lambda$.
  % and $\pr(\kappa(R) >j) \leq (\nu/\lambda)^{j} / j!$ for each positive integer $j$.
\end{theorem}
  %\m{shorten statement?}
  %\m{no point in giving a weight for number of vertices?}
  %\m{could give separate weights for bridges and non-bridge edges}
  %
  %Both this result and the next in fact hold without the assumption that the function $f$ is multiplicative.
%
%  We may extend Theorem~\ref{prop.tauconn} to handle large components,
%  just as Theorem 2.5 extends Theorem 2.2 in~\cite{msw05}.
  %For any graph $G$ and positive integer $t$
%  Let $\kappa_t(G)$ denote the number of components with at least $t$ vertices.
  %The next lenmma will be used in the proof of Lemma~\ref{lem.mc0}.
%\m{wanted?}

%\begin{prop} \label{prop.tauconn-t}
  %Let the finite non-empty weighted set $\ca,\tau$ of graphs be bridge-addable,
  %Let $\lambda>0$, $\nu>0$ and $\tau=(\lambda,\nu)$.
%  If $\ca$ is bridge-addable and $R \in_{\tau} \ca$, and $t$ is a positive integer, then
  %Let $\ca$ be a finite non-empty bridge-addable set of graphs.
  %Let $\lambda>0$, $\nu>0$ and $\tau=(\lambda,\nu)$; and let $R \in_{\tau} \ca$.
%\[ \kappa_t(R) \leq_s 1+ \Po(\nu / \lambda t).\]  %\Po(\frac{\nu}{\lambda t})
  %; and so in particular $\pr(R \mbox{ is connected}) \geq e^{-\nu/\lambda t}$.
%\end{prop}

%\noindent
  %The next result generalises~(\ref{eqn.b-add-frag}).
  % Lemmas 2.5 and 2.6 of~\cite{cmcd09}, and improves on these results even in the case $\tau={\bf 1}$.
  %(Note that we do not assume that $\ca$ is minor-closed.)

\begin{theorem} \label{prop.fragbound}
  %Let the finite non-empty weighted set $\ca,\tau$ of graphs be bridge-addable,
  %Let $\lambda>0$, $\nu>0$ and $\tau=(\lambda,\nu)$.
  %and let $R \in_{\tau} \ca$.
  %Let the weighted graph class $\ca,\tau$ be bridge-addable;
  %Let $\lambda>0$, $\nu>0$ and $\tau=(\lambda,\nu)$;
  %let $n$ be a positive integer such that $\ca_n$ is non-empty;
  %and let $R_n \in_{\tau} \ca$.
  If $\ca$ is finite and bridge-addable and $R \in_{\tau} \ca$, then
\[ \E[\frag(R)] < \frac{2\nu}{\lambda}. \]
\end{theorem}


  %For any class $\ca$ of graphs let $\ca_{n,m}$ be the set of graphs $G$ in $\ca_n$
  %with $e(G)=m$, that is with exactly $m$ edges.
  %For $\lambda>0$, let $A_{n, \lambda} = \sum_{m \geq 0} |\ca_{n,m}| \lambda^m$.
  %We say that $\ca,\lambda$ has growth constant $\gamma$ if
  %$(\frac{A_{n,\lambda}}{n!})^{1/n} \to \gamma$ as $n \to \infty$.
  
  %It would be natural to ask if for each $\tau$ we have
  %$\pr[R_n \mbox{ is connected }] \geq (1+o(1)) e^{-\frac{\nu}{2\lambda_0}}$ as $n \to \infty$.


  %Part (c) of the Corollary? in~\cite{cmcd2012} shows that, in the special case
  %of an addable minor-closed graph class, $\pr[R_n \mbox{ is connected }]$
  %tends to a limit as $n \to \infty$, and identifies this limit as
  %$e^{-C(\rho)} = 1/A(\rho)$.

  Before we introduce the asymptotic results for a general bridge-alterable set of graphs,
  let us record some results on random forests $R_n \in_{\tau} \cf$ which generalise the results
  mentioned earlier for uniform random forests $R_n \inu \cf$ -- 
  see for example~\cite{cmcd-rwg2012} where these results are proved in a general setting.
  Observe that $\tau(F)= f(\bar{K_n}) (\lambda/\nu)^{e(F)} \nu^{n}$ for each $F \in \cf_n$
  (where $\bar{K_n}$ denotes the graph on $[n]$ with no edges):
  thus $\tau(F) \propto (\lambda/\nu)^{e(F)}$,
  and the only aspect of the weighting that matters is the ratio $\lambda/\nu$.
  %
\begin{theorem}\label{prop.asympt-forests}
  %Suppose that $f(G)=1$ for each one-vertex graph $G$.
  Consider $R_n \in_{\tau} \cf$, where $\cf$ is the class of forests.
  Then $\; \kappa(R_n)$ converges in distribution to $1+ \Po(\frac{\nu}{2\lambda})$, so
  $\pr(R_n \mbox{ is connected}) \to e^{-\frac{\nu}{2 \lambda}}$;
  $\E[\kappa(R_n)] \to 1 + \frac{\nu}{2 \lambda}$ as $n \to \infty$; and 
  $\E[ \frag(R_n)] \to \frac{\nu}{\lambda}$ as $n \to \infty$.
 % \m{$var \to \infty$}
\end{theorem}
  %(We do not need two parameters $\lambda$ and $\nu$ here, just their ratio.)
  \noindent
  Now we consider asymptotic results for a bridge-alterable set of graphs.
  These results generalise and amplify inequalities~(\ref{eqn.b-alt}) and~(\ref{eqn.b-alt-frag2});
  and Theorem~\ref{prop.asympt-forests} shows that each of inequalities~(\ref{eqn.asympt-comps})
  to~(\ref{eqn.asympt-frag}) is best-possible for a bridge-alterable class of graphs.
  % \m{except for var?} \m{combine?}
  %
\begin{theorem}\label{prop.asympt-conn}
  Suppose that $\ca$ is bridge-alterable %, $f$ is multiplicative 
  and  $R_n \intau \ca$.  Then
\begin{equation} \label{eqn.asympt-comps}
  \kappa(R_n) \preceq_s 1+ \Po(\frac{\nu}{2\lambda}) \;\; \mbox{ asymptotically},
\end{equation}
  and so in particular
\begin{equation} \label{eqn.asympt-conn}
     \liminf_{n \to \infty} \pr(R_n \mbox{ is connected}) \geq e^{-\frac{\nu}{2\lambda}};
\end{equation}
  and
\begin{equation} \label{eqn.asympt-Ekappa}
 \limsup_{n \to \infty} \E[\kappa(R_n)] \leq 1+ \frac{\nu}{2\lambda}.
\end{equation}
\end{theorem}
  %
\begin{theorem} \label{prop.fragbound2}
  If $\ca$ is bridge-alterable %, $f$ is multiplicative 
  and $R_n \in_{\tau} \ca$, then
\begin{equation} \label{eqn.asympt-frag}
  \limsup_{n \to \infty} \E[\frag(R_n)] \leq \frac{\nu}{\lambda}.
\end{equation}
\end{theorem}
  
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\medskip

  Now consider rooted graphs, starting with rooted forests.  Recall that $\cf^{o}$ denotes the class of rooted forests.
  
\begin{theorem}\label{prop.rooted-forests}
  Consider $R^{o}_n \in_{\tau} \cf^{o}$. % where $\cf$ is the class of forests.
  As $n \to \infty$, 
  $\kappa(R^{o}_n)$ converges in distribution to $1+\Po(\frac{\nu}{\lambda})$; and so
  $\pr(R_n^{o} \mbox{ is connected}) \to e^{-\frac{\nu}{\lambda}}$, and
  $\E[\kappa(R_n^{o})] \to 1 + \frac{\nu}{\lambda}$ as $n \to \infty$.
  In contrast, $\E[ \frag(R_n^{o})] \to \infty$ as $n \to \infty$.
\end{theorem}  
  
    Our final result here is non-asymptotic and may be compared with
  Theorem~\ref{prop.tauconn}.
  It generalises~(\ref{eqn.rooted-conn}) and~(\ref{eqn.rooted-comps}).
  Theorem~\ref{prop.rooted-forests} on rooted forests shows that it is best possible,
  and that there is no rooted-graph analogue for Theorem~\ref{prop.fragbound} (which bounds $\E[ \frag(R_n)]$).

\begin{theorem} \label{prop.rooted}
  Let $\ca$ be finite and bridge-addable, and let $R^{o} \in_{\tau} \cao$.
  Then
\[ \kappa(R^{o}) \leq_s 1+ \Po(\nu/\lambda); \]
  and in particular $\; \pr(R^{o} \mbox{ is connected}) \geq e^{-\nu/\lambda}$, and
  $\E[\kappa(R^{o})] \leq 1+ \nu/\lambda$.
\end{theorem}
  
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\section{Proofs for non-asymptotic results}  
\label{sec.proof-non-a}

  In this section we prove the non-asymptotic results above, namely Theorems~\ref{prop.tauconn},
  \ref{prop.fragbound}, except that we leave proofs for rooted graphs to Section~\ref{sec.rooted}. %and~\ref{prop.rooted}.
  
  Given a graph $G$, let $\Bridge(G)$ denote the set of bridges, and note that
   $|\Bridge(G)|=e_0(G)$;
  and let $\Cross(G)$ denote the set of `non-edges' or `possible edges' between components, and let
  $\cross(G)=|\Cross(G)|$.  %There is one crucial observation.
  % concerning a bridge-addable class $\ca$ of graphs.
  We start with two basic lemmas about graphs.
  The first is just an observation, and needs no proof.
  
\begin{lemma} \label{lem.basic1}
  Let the set $\ca$ of graphs be bridge-addable. 
  If $G \in \ca$ and $e \in \Cross(G)$,
  then the graph $G' = G +e$ obtained from $G$ by adding $e$ is in $\ca$ and $e$ is a bridge of $G'$;
  and $\tau(G) = \tau(G') \cdot (\nu/\lambda)$.
\end{lemma}
  %
  %The above lemma is just an observation, and needs no proof.
  %




  
\begin{lemma} \label{lem.basic2} \cite{msw05}
  If the graph $G$ has $n$ vertices, then $e_0(G) \leq n - \kappa(G)$;
  and if $\kappa(G)=k+1$ then $\cross(G) \geq k(n-k) + \binom{k}{2} \geq k(n-k)$.  
\end{lemma}  

\begin{proof}
  Observe that $\kappa(G) + e_0(G) = \kappa(\tilde{G}) \leq n$, so $e_0(G) \leq n-\kappa(G)$.
  %It is well known that $G$ has at most $n-\kappa(G)$ bridges,
  Now consider the second inequality, and assume that $\kappa(G)=k+1$.
  %Since the number of possible edges between two disjoint sets $X$ and $Y$ is $|X| |Y|$, and
  Since if $0<|X| \leq |Y|$ then $|X| |Y| >(|X|-1)(|Y|+1)$, we see that $\cross(G)$ is minimised when
  $G$ consists of $k$ singleton components and one other component.
\end{proof}  
\medskip

  Now let us recall a well-known elementary fact.
  Let $X$ and $Y$ be random variables taking non-negative integer values, and suppose that $X \leq_s Y$: then
\begin{equation} \label{eqn.exp}
  \E X \leq \E Y.
\end{equation}
  To prove this, note that
\[ \E X = \sum_{t \geq 1} \pr(X \geq t) \leq \sum_{t \geq 1} \pr(Y \geq t) = \E Y. \]


  The next two lemmas concern bounding a random variable by a Poisson-distributed random variable.
  The first lemma is stated in a general form which quickly gives the second and which is 
  suitable also for using later.  %in Section~\ref{sec.asympt}.
\begin{lemma} \label{lem.po1}
  Let the random variable $X$ take non-negative integer values. %in $\{1,2,\ldots\}$.
  Let $\alpha>0$ and let $Y \sim \Po(\alpha)$. Let $k_0$ be a positive integer, and suppose that
\[ \pr(X=k+1) \leq \frac{\alpha}{k+1} \, \pr(X=k) \;\; \mbox{ for each } k=0,1,\ldots,k_0-1. \]
  Then
\[ \pr(k_0 \geq X \geq k) \leq \pr(Y \geq k) \;\; \mbox{ for each } k=0,1,\ldots,k_0 .\]
  %where $Y \sim \Po(\alpha)$.
\end{lemma}
\smallskip

\begin{proof}
  Observe that, for each $k=0, 1, \ldots, k_0-1$ we have
\[ \pr(X=k+i) \leq \frac{\alpha^i}{(k+i)_i} \, \pr(X=k) \;\; \mbox{ for each } i=1,\ldots,k_0 -k. \]
  (Here we are using $(x)_i$ to denote the $i$-term product $x(x-1)\cdots(x-i+1)$.)
  Clearly $\pr(k_0 \geq X \geq 0) \leq 1 = \pr(Y \geq 0)$.
  Let $k_0 > k \geq 0$ and suppose that $\pr(k_0 \geq X \geq k) \leq \pr(Y \geq k)$.
  We want to show that $\pr(k_0 \geq X \geq k+1) \leq \pr(Y \geq k+1)$
  to complete the proof by induction.  This is immediate if
  $\pr(X = k) \geq \pr(Y = k)$, so assume that this is not the case. Then 
 \begin{eqnarray*}
   \pr(k_0 \geq X \geq k+1)
  &=&
    \sum_{i=1}^{k_0-k} \pr(X=k+i) \\
  &\leq &
   \pr(X=k) \sum_{i \geq 1} \frac{\alpha^i}{(k+i)_i} \\
  & \leq &
   \pr(Y=k) \sum_{i \geq 1} \frac{\alpha^i}{(k+i)_i} \\
  & = &
   \pr(Y \geq k+1)
\end{eqnarray*}
  as required.
%  and we have shown that $\kappa(R)$ is stochastically at most $1+ \Po(\nu/\lambda)$.
\end{proof}  
\medskip

  \noindent
  From the last lemma with $k_0$ large we obtain
\begin{lemma} \label{lem.po2} (see~\cite{msw05})
  Let the random variable $X$ take non-negative integer values. %values in $\{1,2,\ldots\}$.
  Let $\alpha>0$ and suppose that
\[ \pr(X=k+1) \leq \frac{\alpha}{k+1} \, \pr(X=k) \;\; \mbox{ for each } k=0,1,2,\ldots \]
  Then $X \leq_s Y$
%\[ \pr(k_0 \geq X \geq k) \leq \pr(Y+1 \geq k) \;\; \mbox{ for each } k=1,\ldots,k_0-1 \]
  where $Y \sim \Po(\alpha)$.
\end{lemma}

\begin{proof}
  Fix $k \geq 0$.  Let $\eps>0$ and choose $k_0 \geq k$ such that $\pr(X>k_0)<\eps$.
  By Lemma~\ref{lem.po1}
\[ \pr(X \geq k) = \pr(k_0 \geq X \geq k) + \pr(X>k_0) \leq \pr(Y \geq k)+ \eps,\]
  and thus $ \pr(X \geq k) \leq \pr(Y \geq k)$.
\end{proof}
\medskip

\begin{proofof} {Theorem~\ref{prop.tauconn}}
  %(We do not assume that $f$ is multiplicative.)
  %Given a graph $G$, let $\Bridge(G)$ denote the set of bridges, so that $|\Bridge(G)|=e_0(G)$;
  %and let $\Cross(G)$ denote the set of non-edges between components, and let
  %$\cross(G)=|\Cross(G)|$.
  %
  %Observe that if $G \in \ca$ and $e \in \Cross(G)$,
  %then the graph $G' = G +e$ obtained from $G$ by adding $e$ is in $\ca$ and $e$ is a bridge of $G'$;
  %and $\tau(G) = \tau(G') \cdot (\nu/\lambda)$.
  %
  %Observe that if $e \in \Bridge(G)$ and $G'$ is the graph $G \setminus e$ obtained from $G$
  %by deleting $e$, then $e \in \Cross(G')$ and $\tau(G) = \tau(G') \cdot (\lambda/\nu)$.  
  %
  It suffices to assume that $\ca$ is $\ca_n$ for some $n$, since the sets
  $\ca_n$ are disjoint. Let $\ca_n^{k}$ denote the set of graphs in $\ca_n$ with $k$ components.
   Let $1 \leq k \leq n-1$.
  %consider $R_n \in_{\tau} \ca$ rather than $R$.
  %As in~\cite{msw05} we use the facts that each $n$-vertex graph $G$ with
  %$\kappa(G)=k$ has at most $n-k$ bridges, and in each $n$-vertex graph $G$
  %with $\kappa(G)=k+1$ there are at least $k(n-k) + {k \choose 2} \geq k(n-k)$
  %\m{prove this here?}
  %non-edges between components. Thus, using also Observation~\ref{obs1},
  By Lemmas~\ref{lem.basic1} and~\ref{lem.basic2}
\begin{eqnarray*}
 \tau(\ca_n^{k}) \cdot(n-k)
 %& \geq & \sum_{G,e} \{ \tau(G): G \in \ca_n^{k}, e \in \Bridge(G)\}\\
 & \geq &
  \sum_{G \in \ca_n^{k},\, e \in \Bridge(G)} \tau(G)\\
  & \geq & \frac{\lambda}{\nu}
  \sum_{G' \in \ca_n^{k\!+\!1},\, e \in \Cross(G')} \tau(G')\\
 %& \geq &  \sum_{H,e} \{ \tau(H): H \in \ca_n^{k+1}, e \in \Cross(H) \} \cdot (\lambda/\nu)\\
 & \geq & \frac{\lambda}{\nu} \
   \tau(\ca_n^{k+1}) \cdot k(n-k) %\cdot (\lambda/\nu).
\end{eqnarray*}
    Therefore
\[   \tau(\ca_n^{k+1}) \leq \frac{\nu}{\lambda k} \  \tau(\ca_n^{k}). \]
%  
%\begin{eqnarray*}
% && \sum_{G} \{ \tau(G): G \in \ca_n, \kappa(G)=k\} \cdot(n-k)\\
% & \geq &
%  \sum_{G,e} \{ \tau(G): G \in \ca_n, \kappa(G)=k, e \in \Bridge(G)\}\\
% & \geq &
%  \sum_{H,e} \{ \tau(H): H \in \ca_n, \kappa(H)=k+1, e \in \Cross(H) \} \cdot (\lambda/\nu)\\
% & \geq &
%   \sum_{H} \{ \tau(H): H \in \ca_n, \kappa(H)=k+1\} \cdot k(n-k) \cdot (\lambda/\nu).
%\end{eqnarray*}
  %
%  Therefore
%\[  \sum_{G} \{ \tau(G): G \in \ca_n, \kappa(G)=k+1\} \leq
%    \frac{\nu}{\lambda k}  \sum_{G} \{ \tau(G): G \in \ca_n, \kappa(G)=k\}, \]
  %
  %Now let $R \in_{\tau} \ca$ and let $X=\kappa(R)-1$.  Then
  Thus for $R \in_{\tau} \ca$
\[
  \pr(\kappa(R) = k+1) \leq \frac{\nu}{\lambda k} \pr(\kappa(R) = k) \;\; \mbox{ for each } k=1,2,\ldots.
\]
  and so, writing $X=\kappa(R)-1$
\[
  \pr(X = k+1) \leq \frac{\nu}{\lambda (k+1)} \pr(X = k) \;\; \mbox{ for each } k=0,1,2,\ldots,
\]
%  and so for $k,i = 1,2,\ldots$
%\begin{equation} \label{eqn.comps}
%  \pr(\kappa(R) = k+i) \leq \frac{(\nu/\lambda)^i}{(k+i-1)_i} \pr(\kappa(R) = k).
%\end{equation}
  %
  %Now %let $X \sim \kappa(R)$,  let $\alpha= \nu/\lambda$ and let $Y \sim\Po(\alpha)$.
  %Then by Lemma~\ref{lem.po2} $X \leq_s Y$.
  Hence if $\alpha= \nu/\lambda$ and $Y \sim\Po(\alpha)$ we have  $X \leq_s Y$ by Lemma~\ref{lem.po2}.
  %
  Finally, %the bound on $\E[\kappa(R)]$ now follows from
  by~(\ref{eqn.exp}), $\; \E[\kappa(R)] = 1+\E[X] \leq 1+ \E[Y] = 1 + \nu/\lambda$.
\end{proofof}
  
  %
%  Then for $k,i = 1,2,\ldots$
%\[ \pr(X=k+i) \leq \frac1{\lambda^i (k+1-i)_{i}} \pr(X=k).\]
 % Clearly $\pr(X \geq 1)=1=\pr(Y+1 \geq 1$.
 % Let $k \geq 1$ and suppose that $\pr(X \geq k) \leq \pr(Y+1 \geq k)$.
 % We want to show that $\pr(X \geq k+1) \leq \pr(Y+1 \geq k+1)$
 % to complete the proof by induction.  This is immediate if
 % $\pr(X = k) \geq \pr(Y+1 = k)$, so assume this is not the case.
 % Then by~(\ref{eqn.comps})
 %\begin{eqnarray*}
 %  \pr(X \geq k+1)
 % &\leq &
 %  \pr(X=k) \sum_{i \geq 1} \frac{(\nu/\lambda)^i}{(k+i-1)_i} \\
 % & \leq &
 %  \pr(Y+1=k) \sum_{i \geq 1} \frac{(\nu/\lambda)^i}{(k+i-1)_i} \\
 % & = &
 %  \pr(Y+1 \geq k+1),
%\end{eqnarray*}
%  and we have shown that $\kappa(R)$ is stochastically at most $1+ \Po(\nu/\lambda)$.
  %To complete the proof,
  %
  %
  %
  %Finally note that %if $X \sim \Po(\alpha)$ then
%\[ \pr(Y \geq j) %= e^{-\alpha} \sum_{k \geq j} \alpha^k /k!
%  = e^{-\alpha} \frac{\alpha^j}{j!} \sum_{i \geq 0} \frac{\alpha^i}{(j+i)_i}
%  \leq \frac{\alpha^j}{j!} \]
  %which completes the proof.
%\end{proofof}
  \bigskip  

%  Following~\cite{msw05} we note that (a)
%  for $G \in \ca_n$ with $\kappa_t(G)=k$, the number of edges $e$ of $G$ with $\kappa_t(G \setminus e)=k+1$
%  is at most $n-k-(k+1)(t-1) \leq n-kt$; \m{include proof here?}
%  and (b) for $G \in \ca_n$ with $\kappa_t(G)=k+1$, the number of non-edges
%  $e$ of $G$ such that $G + e \in \ca_n$ and $\kappa_t(G + e)=k$
%  is at least $(n-kt)kt + {k \choose 2} t^2 \geq (n-kt)kt$.
%  It is now easy to complete the proof by adapting the proof of Theorem~\ref{prop.tauconn}.
%\end{proofof}
%  \bigskip
  
  To prove Theorem~\ref{prop.fragbound} we use two lemmas.  The first is another basic lemma on graphs,
  from~\cite{cmcd08}.  We include the short proof here for completeness.
  %
\begin{lemma} \label{lem.basic3} \cite{cmcd08}
  If the graph $G$ has $n$ vertices, then $\cross(G) \geq (n/2) \cdot \frag(G)$.  
\end{lemma}   
  %
\begin{proof}
  An easy convexity argument shows that if $x,x_1,x_2,\ldots$ are
  positive integers such that each $x_i \leq x$ and $\sum_i x_i=n$
  then $\sum_i \binom{x_i}{2} \leq \frac12 n (x-1)$.
  For, if $n=ax+y$ where $a \geq 0$ and $0 \leq y \leq x-1$ are integers, then
  \[ \sum_i \binom{x_i}{2} \leq a \binom{x}{2} + \binom{y}{2}
  \leq a \binom{x}{2} + \frac{y(x-1)}{2} = \frac12 n (x-1).\]
  %
  Hence if we denote the maximum number of vertices in a component by $x$,
  so that $\frag(G) = n-x$, then 
  %the number of possible edges between components is at least
  \[\cross(G) \geq \binom{n}{2} - \frac12 n (x-1)
  = \frac12 n (n-x) = \frac12 n \ \frag(G)\]
  as required.
  %and thus  $\cross(G) \geq \frac12 n \ \frag(G)$. 
\end{proof}
\medskip
  
  The next lemma is phrased generally so that it can also be used later. % as well as here.
  
\begin{lemma} \label{lem.frag1}
  Let $\ca=\ca_n$ be bridge-addable, and let $R \in_{\tau} \ca$.
  Let $\beta>0$ and assume that $\cross(G) \geq \beta n \cdot \frag(G)$ for each $G \in \ca$.
  Then
\[  \E [\frag(R)] \leq \frac{\nu}{\beta \lambda}.  \]  
\end{lemma}
  
\begin{proof}
  Using Lemma~\ref{lem.basic1} for the second inequality, %and~\ref{lem.basic3}
\begin{eqnarray*}
 \beta n \, \sum_{G \in \ca} \tau(G) \, \frag(G)
 & \leq &
   \sum_{G \in \ca,\, e \in \Cross(G)}  \tau(G) \\
   %\sum_{G, e} \{ \tau(G): G \in \ca, e \in \Cross(G) \} \\
% & \leq &  \sum_{G, e} \{ \lambda^{e(G)} \nu^{\kappa(G)}:G \in \ca_n, e \not\in E(G), G \cup e \in \ca_n\}\\
& \leq &
   \frac{\nu}{\lambda} \ \sum_{G' \in \ca,\, e \in \Bridge(G')} \tau(G')\\
   %\frac{\nu}{\lambda} \ \sum_{G', e} \{ \tau(G'): G' \in \ca, e \in \Bridge(G')\}\\
% & = &
%   (\nu/\lambda) \sum_{G'\in \ca_n} \left( \lambda^{e(G')}\nu^{\kappa(G')}
%   |\{e \in E(G'): e \mbox{ a bridge} \}| \right) \\
 & = &
   \frac{\nu}{\lambda} \ \sum_{G \in \ca} \tau(G) \cdot e_0(G).
   % \;\; \leq \;  (\nu/\lambda) \tau(\ca_n) \cdot cn/2.
\end{eqnarray*}
  Thus
\[
  \E[\frag(R)]  \leq \frac{1}{ \beta n} \, \frac{\nu}{\lambda} \, \E[e_0(R)] <  \frac{\nu}{\beta \lambda}
  %= (\tau(\ca_n))^{-1} \sum_{G \in \ca_n} \lambda^{e(G)} \nu^{\kappa(G)} \frag(G) \leq c\ \nu/\lambda
\]
  as required.
\end{proof}
  \medskip
  
  
\begin{proofof} {Theorem~\ref{prop.fragbound}}
  As in the proof of Theorem~\ref{prop.tauconn}
  it suffices to assume that $\ca$ is $\ca_n$ for some $n$.
  %For each $G \in \ca$ let $\cross(G)$ be the set of non-edges between distinct components of $G$.
  %and let $\add(G)$ be the number of non-edges
  %
  %As noted in~\cite{cmcd08} \m{include proof here?} we have
  %$\cross(G) \geq (n/2) \frag(G)$ for all $n$-vertex graphs. Hence,
  By Lemma~\ref{lem.basic3}, we may now complete the proof using Lemma~\ref{lem.frag1} with $\beta= \frac12$.
\end{proofof}
\bigskip

  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



  
\section{Proofs of asymptotic results} %Theorem~\ref{prop.asympt-conn}}
\label{sec.asympt1}

  In this section we prove the asymptotic results 
  Theorems~\ref{prop.asympt-conn} and~\ref{prop.fragbound2}.
  Assume throughout that $\ca$ is bridge-alterable. % and $f$ is multiplicative.
  %
  Let us focus first on %~(\ref{eqn.asympt-conn}) in 
  Theorem~\ref{prop.asympt-conn}, and in particular on~(\ref{eqn.asympt-comps}).
  The proof goes roughly as follows.
  We first see that it suffices to prove inequality~(\ref{eqn.impliesmain3}) below concerning $\kappa(\RF)$,
  where $\RF$ is a random forest on $[n]$ which we define below, with probabilities
  depending on degrees.
  %, and to prove~(\ref{eqn.impliesmain3}) below concerning $\kappa(\RF)$.
  Then we use a key result from~\cite{amr2011} which tells us about average sizes of components
  of $\RT$ with an edge deleted, where $\RT$ is $\RF$ conditioned on being a tree.
  We find that $\pr(\kappa(\RF)=2)$ is suitably smaller than $\pr(\kappa(\RF)=1)$;
  and from this we deduce that in general $\pr(\kappa(\RF)=k+1)$ is suitably smaller than $\pr(\kappa(\RF)=k)$,
  and so we can use Lemma~\ref{lem.po1}.
 % Theorem~\ref{prop.fragbound2} later.
%  To prove~(\ref{eqn.asympt-conn}), we may fix a bridgeless graph $G \in \ca_W$, and prove that,
%  if $R \intau [G]$ then
%\begin{equation} \label{eqn.impliesmain}
%  \pr({R \mbox{ is connected}}) \geq e^{-\frac{\nu}{2 \lambda_0}+o(1)}
%\end{equation}
%  where $o(1) \rightarrow 0$ as $W \rightarrow \infty$. 

  Now for more details.
  We may define an equivalence relation on graphs by setting $G \sim H$ if $\tilde{G}=\tilde{H}$.
  Let $[G]$ denote the equivalence class of $G$, that is, the set of graphs $H$ such that
  $\tilde{H}=\tilde{G}$.
  Let $W$ be a positive integer.
  Since $\ca$ is bridge-alterable, if $G \in \ca_W$ then $[G] \subseteq \ca_W$.
  Thus $\ca_W$ can be written as a disjoint union of equivalence classes.
  
  To prove~(\ref{eqn.asympt-comps}), %Theorem~\ref{prop.asympt-conn}, 
  we may fix a (large) positive integer $W$,
  a bridgeless graph $G_0 \in \ca_W$, an integer $t \geq 1$ and real $\eps>0$; and prove that,
  if $\RG \intau [G_0]$ then
\begin{equation} \label{eqn.impliesmain2}
  \pr(\kappa(\RG) \geq t+1) \leq \pr(\Po(\frac{\nu}{2\lambda}) \geq t) + \eps
\end{equation}
  if $W$ is sufficiently large. %(We write $\RG$ rather than $R^{G_0}$ for brevity.)
  %Denote $[G_0]$ by $\cb$.
  Since we are now restricting attention to $[G_0]$ %and $f(H)=f(G_0)$ for each $H \in \cb$,
  we may assume that $f(G_0)= 1$.  Denote $\kappa(G_0)$ by $n$: we may assume that $n \geq 2$
  (for otherwise the connected graph $G_0$ is the only graph in $[G_0]$).
  %We will consider~(\ref{eqn.asympt-comps}) and Theorem~\ref{prop.fragbound2} later.

  %Fix a bridgeless graph $G$ on vertex set $\{1,\ldots,W\}$ and let $\cb=[G]$.
  Write $C_1,\ldots,C_n$ for the components of $G_0$, and let $w_i=|V(C_i)|$
  for $i=1,\ldots,n$, so that $W = \sum_{i=1}^n w_i$.
  We use the vector $w=(w_1,\ldots,w_n)$ together with the weighting $\tau$ to define a probability measure on the
  set $\cf_n$ of forests on~$[n]$. %with vertex set $\{1,\ldots,n\}$.
  Given $F \in \cf_n$, let
\[ \mass{F} = \prod_{i=1}^n w_i^{d_F(i)} \cdot \lambda^{e(F)} \nu^{\kappa(F)} \]
  where $d_F(i)$ denotes the degree of vertex $i$ in the forest $F$.
  Also, let $K = \sum_{F \in \cf_n} \mass{F}$, and let $\RF$ be a random element of $\cf_n$ with
  $\pr(\RF =F) = \mass{F}/K$ for each $F \in \cf_n$.
  %We say ${\bf F}$ is {\em distributed according to} ${\bf w}$.
%
  Corresponding to Lemma 2.3 of~\cite{amr2011}, we have
  %we may see that %for ${\bf H} \intau \cb$,
  %$\pr( \RG \mbox{ is connected}) = \pr(\RF \mbox{ is connected})$, and indeed
\begin{equation} \label{eqn.stocheq}
  \kappa(\RG) =_s \kappa(\RF).
\end{equation}
  %To check that indeed~(\ref{eqn.stocheq}) holds,  \bigskip
  %
\begin{proofof}{(\ref{eqn.stocheq})}
    %We construct a flow from $\scr{B}$ to $\cf_n$ in the following fashion:
    Denote $[G_0]$ by $\cb$.
    Given $H \in \cb$, let $g(H)$ be the graph obtained from $H$ by contracting
    each $C_i$ to the single vertex $i$. % for each $i=1,\ldots,n$.
    Then $g(H)\in \cf_n$ and $\kappa(H)= \kappa(g(H))$.  Also,
    for each $F \in \cf_n$, the set $g^{-1}(F)$ has cardinality
    $\prod_{i=1}^n w_i^{d_F(i)}$, and so $\tau(g^{-1}(F))= \mass{F}$.
    %$H \in [G]$ in connected if and only if $f(G)$ is connected, 
   It follows that
\begin{eqnarray*}
   \pr(\kappa(\RG)=k) & = & \frac{\tau(\{H\in\cb : \kappa(H)=k \})}{\tau(\cb)}\\
      & = & \frac{ \sum_{\{F\in\cf_n: \kappa(F)=k \}}\tau(g^{-1}(F))}{\sum_{F\in\cf_n}\tau(g^{-1}(F))} \\
      & = & \frac{ \sum_{ \{ F\in\cf_n: \kappa(F)=k \} } \mass{F} }{K} \\
      & = & \pr(\kappa(\RF)=k)
\end{eqnarray*}
  as required.
\end{proofof}
 % \m{spell out!!}
%\end{lemma}
%  Thus to prove~(\ref{eqn.impliesmain}) we shall prove
%\begin{equation} \label{eqn.impliesmain2}
%  \pr({R \mbox{ is connected}}) \geq e^{-\frac{\nu}{2 \lambda_0}+o(1)}.
%\end{equation}
\smallskip

  Now that we have established~(\ref{eqn.stocheq}), in order to prove~(\ref{eqn.impliesmain2}) we may show that
\begin{equation} \label{eqn.impliesmain3}
  \pr(\kappa(\RF) \geq t+1) \leq \pr(\Po(\frac{\nu}{2\lambda}) \geq t) + \eps
\end{equation}
  if $W$ is sufficiently large.
%It turns out that bounds on $\pr({\bf F}\mbox{ is connected})$ follow from
%bounds on the ratio between $\pr(F\in\cf_{n,2})$ and $\pr(F\in\cf_{n,1})$.

  %For $i=1,\ldots,n$, let $\cf_{n,i}$ be the set of forests in $\cf_n$ with $i$ components.
  %(so $F \in \cf_{n,1}$ precisely if $F$ is connected). %For larger $i$ set $\cf_{n,i}=\emptyset$.
 %
  %Next we define $\varphi(F',F)$ as in~\cite{amr2011} except that we multiply by a factor $\frac{\nu}{\lambda}$.
  %then Lemma 3.1 of~\cite{amr2011} holds exactly as stated.
  Given a graph $H$ on $[n]$, let
\[ \cross_w(H) = \sum_{uv \in \Cross(H)} w_u w_v. \]
  Observe that $\cross_w(H)$ equals %$\add(G)=\add_w(G)$ be
  the sum of $w(C) w(C')$ over the unordered pairs $C$ and $C'$ of
  components of $H$, where $w(C)$ denotes $\sum_{i \in V(C)}w_i$.
  %(Thus $\add(G)=0$ if $G$ is connected.)
  % $c(G)$ be the set of connected components of $G$.
  %Given a forest $F \in \cf_n$ and $T\in c(F)$, le
  For forests $F,F' \in \cf_n$ such that $F$ can be obtained from $F'$ by deleting an edge $uv$,
  %(or equivalently $F'$ can be obtained from $F$ by the addition of an edge $uv$ in $\Cross(F)$);
  observe that $\mass{F'} = \frac{\lambda}{\nu} \cdot \mass{F} \cdot w_uw_v$.
  %Much as in~\cite{amr2011}, we let
  For such $F, F'$ we let
  %Writing $T\neq T' \in c(F)$ as shorthand for $\{\{T,T'\}\subseteq c(F):T\neq T'\}$, we let
\begin{equation}
    \varphi(F',F) = \frac{\nu}{\lambda} \cdot \frac{ \mass {F'} }{\cross_w(F)}. \label{eq:phi_def}
\end{equation}
  For all other pairs $F,F'$, we let $\varphi(F',F)=0$.
  
  For $i=1,\ldots,n$, let $\cf_{n}^{i}$ be the set of forests in $\cf_n$ with $i$ components.
  %By using $\varphi$ in a double-counting argument,
  %As in the proof of Lemma 3.1 of~\cite{amr2011}, we may see that
  For each $F \in \cf_{n}^{i+1}$ we have
\[
  \sum_{F'\in\cf_{n}^{i}}\varphi(F',F) = \frac{\mass{F}}{\cross_w(F)} \sum_{uv \in \Cross(F)} w_uw_v = \mass{F};
\]
  %
%\begin{lemma}\label{lem.mass_eq} \m{old Lemma 3.1}
   %For all positive integers $W$ and all positive integer weight vectors
   %$(w_1,\ldots,w_n)$ with $\sum_{j=1}^n w_j=W$, and
  and thus for each $i=1,\ldots,n-1$
\begin{equation} \label{eqn.sumflow}
 		\sum_{F'\in\cf_{n}^{i}}\sum_{F\in\cf_{n}^{i+1}}\varphi(F',F) =
 		\sum_{F\in \cf_{n}^{i+1}} \mass{F} = K\cdot\pr(\RF \in \cf_{n}^{i+1})
\end{equation}
 as in Lemma 3.1 of~\cite{amr2011}.
%\end{lemma}
      
  %Let $W$ be a positive integer and consider any positive integer weight vector
  %$(w_1,\ldots,w_n)$ with $\sum_i w_i = W$. 
  %We may assume that $n \geq 2$.
  %
 Given a tree $T$ on $[n]$ and an integer $k$ with $1 \leq k \leq \lfloor W/2 \rfloor$,
 let $c(T,k)$ be the number of edges $e$ in $T$ such that $T \setminus e$ has a component with weight~$k$.
 % We call the components of $T-e$ {\em pendant subtrees of $T$}.
  %
   % Recall that $K=\sum_{F \in \cf_n} \mass F$, and 
   % let $K' = \sum_{T \in \cf_{n,1}} \mass{T}=K\cdot \pr(\RF \in \cf_{n,1})$. 
    Let $\RT$ be $\RF$ conditioned on being a tree, so that $\RT$ is a random tree on $[n]$
    with $\pr(\RT =T) \propto \mass{T}$.
%\begin{equation}
%  \label{eqn.rand_tree}
%  \pr(\RT=T) = \frac{\mass{T}}{K'}.
%\end{equation}
 %
   The distribution of $\RT$ is exactly as in~\cite{amr2011} -- 
   the weighting is not relevant here, since $e(T)=n-1$ and $\kappa(T)=1$ are fixed,
   and thus $\pr(\RT = T) \propto \prod_{i=1}^{n} w_i^{d_T(i)}$.
   %
   Hence from Section 4 of~\cite{amr2011} we see that for any $\eta>0$, for $W$ sufficiently large we have
\begin{equation} \label{claim.half}
 \sum_{k \geq 1} \frac{\E{[c(\RT,k)]}}{k(W-k)} \leq (1+\eta) \cdot \frac12.
\end{equation}  
  %
  By~(\ref{eqn.sumflow}) with $i=1$,
  %\m{spell out?}
  corresponding to lemma 4.1 of~\cite{amr2011} we have  
\begin{eqnarray*}
    \pr(\RF \in \cf_{n}^{2}) &=&
      \frac{1}{K} \frac{\nu}{\lambda} \sum_{T \in \cf_n^1} \mass{T} \sum_{e \in T} \frac{1}{\cross_w(T-e)}\\
    &=&
      \pr(\RF \in\cf_{n}^{1}) \ \frac{\nu}{\lambda} \sum_{T \in \cf_n^1} \pr(\RT =T) \sum_{k \geq 1}
      \frac{c(\RT,k)}{k(W-k)}
\end{eqnarray*}
  and so
\begin{equation} \label{eqn.fn2_bound}
      \pr(\RF \in \cf_{n}^{2}) = \pr(\RF \in\cf_{n}^{1})\ \frac{\nu}{\lambda}\ \sum_{k \geq 1}
      \frac{\E{[c(\RT,k)}]}{k(W-k)}.
\end{equation}
   From~(\ref{claim.half}) and~(\ref{eqn.fn2_bound}) we see that
   for all $\eta > 0$, for $W$ sufficiently large, for all
   $w_1,\ldots,w_n$ with $\sum_{j=1}^n w_j=W$,
 \begin{equation} \label{eqn.ratio_1_2}
        \pr(\RF \in \cf_{n}^{2}) \leq (1+\eta) \frac{\nu}{2 \lambda} \ \pr(\RF \in \cf_{n}^{1}).
  \end{equation}
            
  Now we can complete the proof of~(\ref{eqn.impliesmain3})
  (and thus of~(\ref{eqn.asympt-comps}) in Theorem~\ref{prop.asympt-conn}) %(first part!),
  as follows, as in the proof of Claim 2.2 in~\cite{amr2011}.
  The next lemma will allow us to assume that $n$ is large, as well as being useful later.
  %
  %First note that %since $f$ is multiplicative,
  %Lemma 3.2 of~\cite{amr2011} holds if we multiply the right side of the inequality
  %by a factor $\frac{\nu}{\lambda}$:
  Using~(\ref{eqn.sumflow}) and the proof of Lemma 3.2 of~\cite{amr2011} we find:
  %
\begin{lemma}\label{lem.smalln2.5} %\m{old Lemma 3.2}
   %For all positive integers $W$ and all positive integer weight vectors
   %$(w_1,\ldots,w_n)$ with $\sum_{j=1}^n w_j=W$, and
   For each $i=1,\ldots,n-1$
   \begin{equation}\label{eqn.smalln2.5}
       \pr(\RF \in \cf_{n}^{i+1}) \leq \frac{\pr(\RF \in \cf_{n}^{i})}{i} \frac{n}{W} \frac{\nu}{\lambda}.
   \end{equation}
\end{lemma} 
 %
 If $W \geq 2n$ then by the above result and Lemma~\ref{lem.po2},
 $\kappa(\RF) \leq_s 1+ \Po(\frac{\nu}{2 \lambda})$ and so~(\ref{eqn.impliesmain3}) holds.
 %and~(\ref{eqn.asympt-comps}) hold.
 Thus we may assume from now on %in our proof of Theorem~\ref{prop.asympt-conn}
 that $W<2n$.
 
 Next we introduce Lemma 3.3 of~\cite{amr2011}.
 For each finite non-empty set $V$ of positive integers,
  let $\cg_V$ denote the set of all graphs on the vertex set $V$,
  %let $\scr{G}^k$ be the set of all graphs with $k$ components and
  %with vertices drawn from $\N$,
  and let $\cg^k_V$ denote the set of all graphs in $\cg_V$ with exactly
  $k$ components.
  %Also, write $\cg_n$ for $\cg(\{1,\ldots,n\})$, and $\cg_n^k$
  %for $\cg^k(\{1,\ldots,n\})$.
  %
  For each positive integer $n$, let $\mu_n$ be a measure on the set of all graphs
  with vertex set a subset of $[n]$ which is \emph{multiplicative on components};
  that is, if $G$ has components $H_1,\ldots,H_k$ then
  $\mu_n(G)=\prod_{i=1}^k \mu_n(H_i)$. 
  Observe that we obtain such a measure
  %on the graphs with vertex sets a subset of $[n]$ which is multiplicative on components, 
  if we set $\mu_n(G) = \mass{G}$ when $G$ is a forest and $\mu_n(G)=0$ otherwise. 
  %
\begin{lemma}\label{lem.connect_1} (\cite{amr2011}) %\m{old Lemma 3.3}
  Suppose there exist $\alpha > 0$ and integers $n \geq m_0 \geq 1$ such that
  \begin{equation}\label{eq.connect_1_a}
    \mu_n(\cg^2_V) \leq \alpha \mu_n(\cg^1_V) \;\;\; \mbox{for all }
    V \subseteq \{1,\ldots,n\} \mbox{ with } |V| \geq m_0.
  \end{equation}
  Then for all integers $k \geq 1$ and $n \geq km_0$
  \begin{equation}
    \label{eq:connect_1}
    \mu_{n}(\cg_{[n]}^{k+1})\leq \frac{\alpha}{k}\mu_n(\cg_{[n]}^k).
  \end{equation}
\end{lemma}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

  We may now complete the proof of~(\ref{eqn.impliesmain3}). % Theorem~\ref{prop.asympt-conn}.
%\begin{proof}[Proof of Claim \ref{cla.impliesmain}] 
% assuming Lemma \ref{lem.ratio_1_2}]
  %
  %Fix $\eta$ with $0 < \eta < 1$, and
  Fix $j \geq t$ large enough that
\[ \sum_{i \geq j} \left(\frac{\nu}{\lambda}\right)^i \frac{1}{i!} \leq \frac{\eps}{2}.\]
  % $2/j! \leq \eta/2$.
  %Let $X= \kappa(\RF)$. \m{use $X$?}
  Fix $\eta > 0$ small enough that, with $\alpha=(1+\eta) \frac{\nu}{2 \lambda}$,
  we have
\[ \pr(\Po(\alpha) \geq t) \leq \pr(\Po(\frac{\nu}{2 \lambda}) \geq t) + \eps/2. \]
  %
  %$W$ big gives $n$ big, so we can use Lemma \ref{lem.connect_1}. \bigskip
  %
  By~(\ref{eqn.ratio_1_2}) and Lemma \ref{lem.connect_1}
  % with $\beta=(1+\eps)\frac12 \frac{\nu}{\lambda}$.
  %(Lemma \ref{lem.ratio_1_2} guarantees that there exists $m_0>0$ such
  %that the hypotheses of Lemma~\ref{lem.connect_1} hold with this choice of $m_0$ and $\beta$).
  %\m{expand?}
  it follows that, for $W$ large enough (recall that $n >W/2$), for all $i$ with $1\leq i \leq j$ we have
%\begin{equation} \label{eqn.krf}
\[  \pr(\kappa(\RF)=i+1) \le \frac{\alpha}{i} \pr(\kappa(R^f)=i). \]
%\end{equation}
  In terms of $X= \kappa(\RF)-1$, this says that
\begin{equation} \label{eqn.krf}
  \pr(X=i+1) \le \frac{\alpha}{i+1} \pr(X=i) \;\; \mbox{ for } i=0,1,\ldots,j-1.
\end{equation} 
 %\[ \pr(R^f \in \cf_{n,i+1}) \leq (1+\eps)^i\frac{ \pr( R^f \in \cf_{n,1})} {2^i i!}.\]
%  Furthermore, writing $\kappa(F)$ for the number of connected components of $F$,
%
%\begin{eqnarray}
%  1 &=& \sum_{i=0}^{n-1}\pr(R^f \in \cf_{n,i+1}) \nonumber\\
%&\leq&  (1+\eps)^{j}\sum_{i=0}^{j-1}\frac{\pr(R^f\in \cf_{n,1})}{2^i i!}
%  + \pr(\kappa(R^f) \geq j+1).\label{eq:impliesmain1}
%\end{eqnarray}
  Also, by Lemma \ref{lem.smalln2.5}, for all $i \geq 1$ %and so (a fortiori) for all $i \geq j$,
%\begin{equation} \label{eqn.krf}
\[ \pr(\RF \in \cf_{n}^{i+1})\leq \left(\frac{n \nu}{W \lambda}\right)^i \frac{1}{i!}
   \leq \left(\frac{\nu}{\lambda}\right)^i \frac{1}{i!}\]
%\end{equation}
  and so it follows by our choice of $j$ that
\[ \pr(X \geq j) = \pr(\kappa(\RF) \geq j+1) \leq \eps/2. \]
  Hence by~(\ref{eqn.krf}) and Lemma~\ref{lem.po1},
\begin{eqnarray*}
 \pr(X \geq t)
 & = & \pr(j-1 \geq X \geq t) + \pr(X \geq j)\\
 & \leq & \pr(\Po(\alpha) \geq t) + \eps/2\\
  & \leq & \pr(\Po(\frac{\nu}{2\lambda}) \geq t) + \eps.
\end{eqnarray*}
%
%  Combining the latter equation with (\ref{eq.impliesmain1}) yields that
%\[  1 \leq (1+\eps)^{j}e^{\frac12}\pr(R^f \in \cf_{n,1}) + \alpha/2,\nonumber\] so
%\begin{equation}\label{eq.implies_fcon}
%  \pr(R^f \in \cf_{n,1}) \geq
%  \frac{1-\alpha/2}{(1+\eps)^{j}e^{\frac12}} \geq \frac{1-\alpha}{e^{\frac12}}.
%\end{equation}
%  As $\alpha > 0$ was arbitrary, (\ref{eq.implies_fcon}) implies that
%  $\pr(R^f \mbox{ is connected}) \geq e^{-\frac12+o(1)}$ which combined with
%  Lemma~\ref{eq.hf_con} proves Claim~\ref{cla.impliesmain}.
%\hfill\qed
%\end{proof}

  %A simpler version of the above argument lets us deduce directly from
  %inequality~(\ref{eq.smalln2.5}) in Lemma~\ref{lem.smalln2.5}
  %the non-asymptotic result, that $\pr(R^f \mbox{ is connected})> e^{-n/W}$
  %for all positive integers $W$ and all weight vectors
  %$(w_1,\ldots,w_n)$ with $\sum_{j=1}^n w_j=W$.


  This completes the proof of~(\ref{eqn.impliesmain3}), and thus of~(\ref{eqn.impliesmain2}) and so
  of~(\ref{eqn.asympt-comps}).
  The inequality~(\ref{eqn.asympt-conn}) follows directly from~(\ref{eqn.asympt-comps}),
  %(just as~(\ref{eqn.b-a-conn}) followed from~(\ref{eqn.b-a-comps})).
  %The case $t=1$ of~(\ref{eqn.asympt-comps}) immediately gives~(\ref{eqn.asympt-conn}).
  so it remains only to prove~(\ref{eqn.asympt-Ekappa}).  Let $\eps>0$.  By Theorem~\ref{prop.tauconn},
  if $Y \sim \Po(\nu/\lambda)$ then
\[ \E[\kappa(R_n) {\bf 1}_{\kappa(R_n) \geq t+1}] \leq \E[(1+Y) {\bf 1}_{Y \geq t}] < \eps/2\]
  if $t$ is sufficiently large.  Fix such a $t$.
  %Now let $\alpha=\frac{\nu}{2 \lambda}$.
  
  Since $\eps$ and $t$ are fixed, by~(\ref{eqn.impliesmain2}) 
  (applied for each value $i=1,\ldots,t$ and with $\eps$ replaced by $\frac{\eps}{2t}$)
  there is an $n_0$ such that for each
  $n \geq n_0$ and each bridgeless graph $G \in \ca_n$, for $R^G \in_{\tau} [G]$ we have
\[ \pr(\kappa(R^G) \geq i+1) \leq \pr(\Po(\frac{\nu}{2\lambda}) \geq i) + \frac{\eps}{2t} \]  
  for each $i=1,\ldots,t$.  It follows that these inequalities hold also with $R^G$ replaced by $R_n$,
  where $R_n \in_{\tau} \ca$ (since $\ca_n$ is a disjoint union of equivalence classes).
  Thus for $n \geq n_0$  
  %(applied for each value up to~$t$, and with $\eps$ replaced by $\frac{\eps}{2t}$), for $n$ sufficiently large
\begin{eqnarray*}
  \E[\kappa(R_n) {\bf 1}_{\kappa(R_n) \leq t}]
  &=& %\sum_{i=1}^{t} i \, \pr(\kappa(R_n)=i) \;\; \leq \;\; 
  \sum_{i=1}^{t} \pr(t \geq \kappa(R_n)\geq i)\\
  & \leq & \sum_{i=0}^{t-1} \pr(\Po(\frac{\nu}{2 \lambda})\geq i) + \frac{\eps}{2}\\
  & \leq & 1+ \E[\Po(\frac{\nu}{2 \lambda})] + \frac{\eps}{2} \; = \; 1 + \frac{\nu}{2 \lambda} + \frac{\eps}{2}.
\end{eqnarray*}  
  Hence, for $n$ sufficiently large, $\E[\kappa(R_n)] \leq 1 + \frac{\nu}{2 \lambda}+ \eps$, and we are done.
   We have now completed the proof of Theorem~\ref{prop.asympt-conn}.



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%




\subsection{Proof of Theorem~\ref{prop.fragbound2}}
\label{sec.fragproof}

  Let $\eps>0$, and let $\ca'_n = \{G \in \ca_n:\frag(G) \leq \eps n\}$.  Then $\ca'_n$ is bridge-addable.
  Also, for each $G \in \ca'_n$ we have $\cross(G) \geq (1-\eps)n \cdot \frag(G)$.
  Hence by Lemma~\ref{lem.frag1} with $\beta=1-\eps$ we have
\[  \E[\frag(R_n) {\bf 1}_{\frag(R_n) \leq \eps n}] < (1-\eps)^{-1} (\nu/\lambda). \]
  Thus it suffices for us to show that
\begin{equation} \label{eqn.frageps}
  \E[\frag(R_n) {\bf 1}_{\frag(R_n) > \eps n}] \; = \; o(1) \;\; \mbox{  as } n \to \infty.
\end{equation}
  
  We proceed as in the proof of Theorem~\ref{prop.asympt-conn}. %~\ref{prop.tauconn}.
  For a graph $G$ on $[n]$ let $\wfrag(G)$ denote $W$ minus the maximum weight $w(C)$ of a component $C$ of~$G$.
  Then corresponding to~(\ref{eqn.stocheq}) we have
\[ \frag (\RG) =_s \wfrag(\RF). \]
  To see this we may argue as in the proof of~(\ref{eqn.stocheq}), %Theorem~\ref{prop.tauconn},
  recalling that for each $H \in \cb =[G_0]$, $g(H)$ is the forest obtained by contracting the components $C_i$ of $G_0$, and  
  noting that we have $\frag(H)=\wfrag(g(H))$.
  Thus it suffices to consider $\RF$
  and show that $\E[X {\bf 1}_{X > \eps W}]$ is $o(1)$ as $W \to \infty$,
  where $X = \wfrag(\RF)$. 
  %and $\wfrag(G)$ denotes $W$ minus the maximum weight of a component of $G$.
  %the weight of $\Frag(G)$.
  \smallskip

  %Denote the ratio $\nu / \lambda$ by $\alpha$.
  Define $\ttau(F)$ to be $\tau(g^{-1}(F))$ for each $F \in \cf_n$; and for $\ca \subseteq \cf_n$ let
  $\ttau(\ca) = \sum_{F \in \ca} \ttau(F) = \tau(g^{-1}(\ca))$.
  Then $\ttau(\cf_n)=\tau(\cb)$, and $\ttau(\ct_n)=\tau(\cc)$ where $\cc$ is the set of connected graphs in $\cb$.
  Thus by Theorem~\ref{prop.tauconn} we have
  $\ttau(\ct_n) \geq e^{-\alpha} \ttau(\cf_n)$, where we let $\alpha$ denote $\nu / \lambda$.
  Recall that we take $f(G) \equiv 1$ without loss of generality, and so
  $\tau(H)= \alpha \lambda^n$ for each connected graph $H$ in $\cb$.
  Also
\[ |g^{-1}(\ct_n)| = \prod_{i=1}^{n} w_i \cdot W^{n-2}\]
   where $W=\sum_{i=1}^n w_i$. % (we take $f(G) \equiv 1$).
   This counting result goes back to Moon~\cite{moon67} in 1967 and R\'enyi~\cite{renyi1970} in 1970
   % (see problem 4.3 of Lov\'asz~\cite{lovaszbook})
   (see also Pitman~\cite{pitman99}) and appears in %the formula for $K'$ in
   the proof of Lemma 4.2 of~\cite{amr2011}.  Hence
\[ \ttau(\ct_n) = \alpha \lambda^n \prod_{i=1}^{n} w_i \cdot W^{n-2}.\]
%\[ \ttau(\ct_n) = \sum_{T \in \ct_n} \mass{T} = \alpha \lambda^n \prod_{i=1}^{n} w_i \cdot W^{n-2}\]
  For each non-empty set $A$ of positive integers, let $\ct_A$ and $\cf_A$ denote respectively the sets of
  trees and forests on vertex set $A$, and let $W_A$ denote $\sum_{i \in A} w_i$.
  Let $1 \leq a \leq n-1$ and let $A \subseteq [n]$ with $|A|=a$.  Denote $[n] \setminus A$ by $\bar{A}$. Then
\begin{eqnarray*}
    \ttau(\cf_A) \ttau(\cf_{\bar{A}})
    & \leq & e^{2 \alpha}  \ttau(\ct_A) \ttau(\ct_{\bar{A}})\\
    &=& \alpha^2 \lambda^n e^{2 \alpha} (\prod_{i=1}^{n} w_i ) \cdot W_A^{a-2} (W-W_A)^{n-a-2}\\
    &=& \alpha^2 \lambda^n e^{2 \alpha} (\prod_{i=1}^{n} w_i) \cdot (W_A(W-W_A))^{-2} W_A^{a} (W-W_A)^{n-a}.
\end{eqnarray*}
  But $x^a(W-x)^{n-a}$ is maximised at $x=\frac{a}{n}W$, so
\[ W_A^{a} (W-W_A)^{n-a} \leq (\frac{a}{n} W)^{a} (\frac{n-a}{n} W)^{n-a} = a^a (n-a)^{n-a} \left(\frac{W}{n} \right)^n.\]
  Thus
\[  \ttau(\cf_A) \ttau(\cf_{\bar{A}}) \leq \alpha^2 \lambda^n e^{2 \alpha} \prod_{i=1}^{n} w_i \cdot (W_A(W-W_A))^{-2} a^a (n-a)^{n-a} \left(\frac{W}{n} \right)^n.\]
  Now by Stirling's formula, there are positive constants $c_1$ and $c_2$ such that for each positive integer $k$
%\begin{equation} \label{eqn.stir}
\[  c_1 k^{k+\frac12} e^{-k} \leq k! \leq c_2 k^{k+\frac12} e^{-k}. \]
%\end{equation}
  Thus
\begin{eqnarray*}
  \sum_{a=1}^{n-1} \binom{n}{a} a^{a} (n-a)^{n-a}
  &=& n! \sum_{a=1}^{n-1} \frac{a^a}{a!} \frac{(n-a)^{n-a}}{(n-a)!}\\
  & \leq & c_1^{-2} n! \, e^n \sum_{a=1}^{n-1} (a(n-a))^{-\frac12}.
\end{eqnarray*}
  But the sum in this last expression is $O(1)$, so 
%\begin{eqnarray*} 
%  \sum_{a=1}^{n-1} (a(n-a))^{-\frac12} & \sim &
%  \int_{1}^{n-1} (x(n-x))^{-\frac12} dx \leq 2 \int_{1}^{\frac{n}{2}} (x(n-x))^{-\frac12} dx \\
%  & \leq & 2 (\frac{2}{n})^{\frac12} \int_{0}^{\frac{n}{2}} x^{-\frac12} dx = 4.
%\end{eqnarray*}
  %
\[  \sum_{a=1}^{n-1} \binom{n}{a} a^{a} (n-a)^{n-a} \; \leq \; c_3 n! e^n \; \leq \; c_4 n^{n+\frac12} \]
  for some constants $c_3$ and $c_4$.
  It follows that
\[ \sum_{ A \subseteq [n]} (W_A W_{\bar{A}})^2 \ \ttau(\cf_A) \ttau(\cf_{\bar{A}})
   \leq \; c_5 \alpha^2 \lambda^n e^{2 \alpha} (\prod_{i=1}^{n} w_i) W^n n^{\frac12} \; 
   = \; c_6 \, \ttau(\ct_n) W^2 n^{\frac12} \]
%
%\[ \sum_{ A \subseteq [n]} (W_A W_{\bar{A}})^2 \ \ttau(\cf_A) \ttau(\cf_{\bar{A}}) \hspace{1.75in} \]
%\vspace{-.2in}
%\[ \hspace{1in}  \leq \;
%    c_5 \alpha^2 \lambda^n e^{2 \alpha} (\prod_{i=1}^{n} w_i) W^n n^{\frac12} \; = \; c_6 \, \ttau(\ct_n) W^2 n^{\frac12} \]
   %
  for some constant $c_5$, and $c_6 = c_6(\alpha) = c_5 \alpha e^{2 \alpha}$.
  %
  Let us introduce a piece of notation: for $z>0$ let %denote by $\sum_{W_A \geq z}$ the sum \[ \sum_{W_A \geq z} s_A
\[ \sum_{z \leq W_A \leq W/2} s_A := \sum \{ W_A \ \ttau(\cf_A) \ttau(\cf_{\bar{A}}): A \subseteq [n], z \leq W_A \leq W/2\}. \]
  Then
%\begin{eqnarray*} 
 %&& \sum \{ w_A \tau(\cf_A) \tau(\cf_{\bar{A}}): A \subseteq [n], z \leq W_A \leq W/2\}\\
 %& \leq & 
\begin{equation} \label{eqn.end}
  \sum_{z \leq W_A \leq W/2} s_A \leq  z^{-1} (\frac{W}{2})^{-2} \sum_{ A \subseteq [n]} (W_A W_{\bar{A}})^2 \ 
  \ttau(\cf_A) \ttau(\cf_{\bar{A}}) 
  %\leq  c_6 (\prod_{i=1}^{n} w_i) \ z^{-1} W^{-2} n^{\frac12} W^n 
  \leq 4 c_6 \ttau(\ct_n) z^{-1} n^{\frac12}.
\end{equation}

% \begin{eqnarray} %\label{eqn.end}
%  \sum_{z \leq W_A \leq \frac{W}{2}} s_A 
%  & \leq & z^{\!-\!1} (\!\frac{W}{2}\!)^{\!-\!2} \!\sum_{ A \subseteq [n]} (W_A W_{\bar{A}})^2 \ 
%  \ttau(\cf_A) \ttau(\cf_{\bar{A}}) \nonumber \\ 
%  %\leq  c_6 (\prod_{i=1}^{n} w_i) \ z^{-1} W^{-2} n^{\frac12} W^n 
%  & \leq & 4 c_6 \ttau(\ct_n) z^{-1} n^{\frac12}. 
%\end{eqnarray}

%\begin{eqnarray} \label{eqn.end}
%  \sum_{W_A \geq z} & \leq & z^{-1} (\frac{W}{2})^{-2} \sum_{ A \subseteq [n]} (W_A W_{\bar{A}})^2 \ 
%  \tau(\cf_A) \tau(\cf_{\bar{A}}) \nonumber \\ 
%  %\leq  c_6 (\prod_{i=1}^{n} w_i) \ z^{-1} W^{-2} n^{\frac12} W^n 
%  & \leq & 4 c_5 \tau(\ct_n) z^{-1} n^{\frac12}.
%\end{eqnarray}
  Now we argue as in the proof of Proposition 5.2 of~\cite{cmcd08}.
  Consider a disconnected graph $G$ on $[n]$, and denote $\wfrag(G)$ by $z$.
  We claim that there is a union of components with weight in the interval $[z/2, W/2]$.
  To see this let $b=W-z$, so that $b$ is the biggest weight of a component.
  Note that $\lfloor \frac{W+b}{2} \rfloor - \lceil \frac{W-b}{2} \rceil +1 \geq b$,
  and so there are at least $b$ integers in the list
  $\lceil \frac{W-b}{2} \rceil,\ldots,\lfloor \frac{W+b}{2} \rfloor$.
 %  
  %interval $[\lceil \frac{W-b}{2} \rceil , \lfloor \frac{W+b}{2} \rfloor]$.
  Thus by considering adding components one at a time we see that there is a union of components, with vertex $A$ say,
  such that $W_A$ is in this set.  Then $W_A \geq \lceil \frac{W-b}{2} \rceil \geq z/2$ and
  $W-W_A \geq W - \lfloor \frac{W+b}{2} \rfloor \geq z/2$.  Thus $A$ or $\bar{A}$ is as required.
  
  From the above, there is an injection from the set of forests $F \in \cf_n$ with $\wfrag(F) \geq z$
  to the set of triples $A, F_A, F_{\bar{A}}$ where
  %$A \subseteq [n]$, $z/2 \leq \wfrag(F)/2 \leq W_A \leq W/2$,
  %$F_A \in \cf_A$, $F_{\bar{A}} \in \cf_{\bar{A}}$
\[ A \subseteq [n], \; z/2 \leq \wfrag(F)/2 \leq W_A \leq W/2, \; F_A \in \cf_A, \; F_{\bar{A}} \in \cf_{\bar{A}} \:
  \mbox{ and } \: \ttau(F)= \ttau(F_A) \ttau(F_{\bar{A}}).
\]
 %  and where $\, \ttau(F)= \ttau(F_A) \ttau(F_{\bar{A}})$.
  It follows that
%\[ \ttau(\cf_n) \, \E[X 1_{X \geq z}] = \sum_{F \in \cf_n: {\wfrag}(F) \geq z} \ttau(F) \wfrag(F)
%   \leq 2 \sum_{W_A \geq z/2} s_A. \] 
\begin{eqnarray*}
  \ttau(\cf_n) \, \E[X 1_{X \geq z}] &=& 
     \sum_{F \in \cf_n: {\wfrag}(F) \geq z} \ttau(F) \, \wfrag(F)\\
  & \leq &
     \!\!\sum_{A \subseteq [n], \frac{z}{2} \leq W_A \leq \frac{W}{2}} \;
     \sum_{F_A \in \cf_A} \sum_{F_{\bar{A}} \in \cf_{\bar{A}}}
     2 W_A \, \ttau(F_A) \, \ttau(F_{\bar{A}})\\
  & = &  \;  2 \!\sum_{\frac{z}{2} \leq W_A \leq \frac{W}{2}} s_A.
\end{eqnarray*} 
  Hence
\[ \E[X 1_{X > \eps W}] \leq \frac{2}{\ttau(\cf_n)} \sum_{\eps W/2 \leq W_A \leq W/2} s_A =O(W^{-\frac12}) \]
  by~(\ref{eqn.end}), and the proof of Theorem~\ref{prop.fragbound2} is complete.




%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



\section{Proofs for rooted graphs}
\label{sec.rooted}

  In this section we prove Theorems~\ref{prop.rooted-forests} and~\ref{prop.rooted} and on rooted graphs.
  First we prove Theorem~\ref{prop.rooted},
  following the lines of the proof of Theorem~\ref{prop.tauconn}.
  \smallskip
  
  \begin{proofof} {Theorem~\ref{prop.rooted}}
  As before, it suffices to assume that $\ca$ is $\ca_n$ for some $n$.
  %since the sets $\ca_n$ are disjoint. 
  Let $1\leq k \leq n-1$.
  Let $\cp$ be the set of pairs $(G^o,e)$ where $G^o$ is a rooted graph on $[n]$ with $k$ components and $e$ is a bridge
  in $G^{o}$ (which we may think of as being oriented towards the root of the component).
  Let $\cq$ be the set of pairs $(H^{o},uv)$, where $H^o$ is a rooted graph on $[n]$ with $k+1$ components, and $uv$ is an
  ordered pair of vertices such that $u$ is the root of its components and $v$ is in a different component.

  There is a natural bijection between $\cp$ and $\cq$.  Given $(G^o,e) \in \cp$, if $u$ is
  the end of $e$ further from the root, we delete the edge $e$ and make $u$ a new root: given $(H^{o},uv) \in \cq$,
  we add an edge between $u$ and $v$ and no longer have $u$ as a root.
  %
  Further, if the $k+1$ components of $H^{o}$ have $n_1,\ldots, n_{k+1}$ vertices respectively,
  then the number of pairs $uv$ such that $(H^{o},uv) \in \cq$ is
  $\sum_{i=1}^{k+1} (n-n_i) = kn$. 
  
  Now much as in the proof of Theorem~\ref{prop.tauconn} we have  
\begin{eqnarray*}
   \tau(\ca_n^{k \, o}) \cdot(n\!-\!k)
    & \geq &
  \sum_{(G^{o},e) \in \cp}  \tau(G^{o}) \; \geq \; \frac{\lambda}{\nu}
  \sum_{(H^{o},uv) \in \cq} \tau(H^{o})\\
  & = & \frac{\lambda}{\nu} \
    \tau(\ca_n^{k+1 \, o}) \cdot k n.
\end{eqnarray*}
  Therefore
\begin{equation} \label{eqn.rooted}
   \tau(\ca_n^{k\!+\!1 \, o}) \leq
   \frac{n\!-\!k}{n} \frac{\nu}{\lambda k} \ \tau(\ca_n^{k \, o}).
\end{equation}
  %
%\begin{eqnarray*}
% && \sum_{G^{o}} \{ \tau(G^{o}): G^{o} \in \cao_n, \kappa(G^{o})=k\} \cdot(n-k)\\
% & \geq &
%  \sum_{G^{o},e} \{ \tau(G^{o}): G^{o} \in \cao_n, (G^{o},e) \in \cp \}\\
% & \geq &
%  \sum_{H^{o},uv} \{ \tau(H^{o}): H^{o} \in \cao_n, (H^{o},uv) \in \cq  \} \cdot (\lambda/\nu)\\
% & = &
%   \sum_{H^{o}} \{ \tau(H^{o}): H^{o} \in \cao_n, \kappa(H^{o})=k+1\} \cdot k n \cdot (\lambda/\nu).
%\end{eqnarray*}
%  Therefore
%\begin{equation} \label{eqn.rooted}
%   \sum_{G^{o} \in \cao_n} \{ \tau(G^{o}): \kappa(G^{o})=k+1\} \leq
%   \frac{n-k}{n} \frac{\nu}{\lambda k} \sum_{G^{o} \in \cao_n} \{ \tau(G^{o}): \kappa(G^{o})=k\}
%\end{equation}
  Thus for $R^{o} \in_{\tau} \cao$
\[
  \pr(\kappa(R^{o}) = k+1) \leq \frac{\nu}{\lambda k} \pr(\kappa(R^{o}) = k) \;\; \mbox{ for each } k=1,2,\ldots,  \]
  and we may complete the proof as for Theorem~\ref{prop.tauconn}.
\end{proofof}
  \bigskip
  
\begin{proofof}{Theorem~\ref{prop.rooted-forests}}
  Consider rooted forests %as in Theorem~\ref{prop.rooted-forests},
  and let $R^{o}_n \in_{\tau} \cf^{o}$.
  Then the first two inequalities above hold at equality, so
\[  
  \pr(\kappa(R^{o}_n) = k+1) = \frac{n-k}{n} \frac{\nu}{\lambda k} \pr(\kappa(R^{o}_n) = k) \;\; \mbox{ for each } k=1,2,\ldots.
\]
  Hence $\kappa(R^{o}_n)$ converges in distribution to $1+\Po(\frac{\nu}{\lambda})$ as $n \to \infty$. 
  Thus to complete the proof of Theorem~\ref{prop.rooted-forests}, it remains only to show that
  $\E[\frag(R_n^{o})] \to \infty$ as $n \to \infty$.
  By what we have just proved and the fact that $|\ct_n^{o}|=n^{n-1}$ %\m{ref needed?}
\[ \tau(\cf_n^{o}) \sim e^{\nu/\lambda} \ \tau(\ct_n^{o}) = \nu e^{\nu/\lambda} (\lambda n)^{n-1}. \] 
  We now obtain a lower bound on $\E[\frag(R_n^{o})]$ by considering forests with two components.
  With sums over say $\log n < j < n/2$, and using Stirling's formula, we have
\begin{eqnarray*}
  \E[\frag(R_n^{o})] & \geq & \tau(\cf_n^{o})^{-1} \cdot \sum_j \binom{n}{j} j \ \tau(\ct_j^{o}) \tau(\ct_{n-j}^{o})\\
  %& \sim &
  %  \left(\!\nu e^{\nu/\lambda}\!\right)^{\!-\!1} (\lambda n)^{\!-\!(n\!-\!1)} n!
  %  \sum_j j \frac{(\lambda j)^{\!j\!-\!1}}{j!} \frac{(\lambda(n\!-\!j))^{\!n\!-\!j\!-\!1}}{(n\!-\!j)!}\\ 
  & \sim &
    \left(\!\nu e^{\nu/\lambda}\!\right)^{-1} (\lambda n)^{-(n-1)} n!
    \sum_j j \frac{(\lambda j)^{j-1}}{j!} \frac{(\lambda(n\!-\!j))^{n-j-1}}{(n\!-\!j)!}\\ 
  &=& \Theta(1) \cdot \frac{n!}{n^{n-1}} \sum_j \frac{j^j}{j!} \frac{(n-j)^{n-j-1}}{(n-j)!}\\
  &=& \Theta(1) \cdot n^{\frac32} e^{-n} \sum_j j^{-\frac12} e^j (n-j)^{-\frac32} e^{n-j}\\
  &=& \Theta(1) \cdot \sum_j j^{-\frac12} \; = \; \Theta(n^{\frac12}),
\end{eqnarray*}  
%\m{use $C'(\rho)=\infty$?}
  and we are done.
\end{proofof}

  \bigskip
  
  
  \noindent
  {\em Aside on the unrooted case}
  
  From the inequality~(\ref{eqn.rooted}) above we may quickly deduce the result in Theorem~\ref{prop.tauconn} that,
  for $R \in_{\tau} \ca$ (where $\ca$ is finite and bridge-addable and not rooted) we have
  $\; \pr(R \mbox{ is connected}) \geq e^{-\nu/\lambda}$.  
  To see this note that we may assume as usual that $\ca$ is $\ca_n$,
  and note also that each graph $G \in \ca_n$ with $\kappa(G)=k+1$
  yields at least $n-k$ rooted graphs in $\cao_n$. 
  Now let $\cc_n$ be the set of connected graphs in $\ca_n$, and use~(\ref{eqn.rooted}) once and then $k-1$ further times:
  we find
\begin{eqnarray*} 
  \tau(\ca_n^{k+1}) 
  &\leq &
     \frac1{n-k}  \tau(\ca_n^{k\!+\!1 \, o}) 
   \; \leq \;
    \frac1{n} \frac{\nu}{\lambda k} \tau(\ca_n^{k \, o}) \\ 
   &\leq &
    \frac1{n} \left(\frac{\nu}{\lambda}\right)^k \frac1{k!} \tau(\cc_n^{o}) \; = \;
   \left(\frac{\nu}{\lambda}\right)^k \frac1{k!} \tau(\cc_n).
\end{eqnarray*}
  Thus
\[ \tau(\ca_n) \leq \sum_{k \geq 0} \left(\frac{\nu}{\lambda}\right)^k \frac1{k!} \tau(\cc_n) = e^{\nu/\lambda} \tau(\cc_n),\]
  and the proof is complete.
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%




\section{Concluding remarks}
\label{sec.concl}

  %As has been noted before, the class $\cf$ of forests seems to be the `least connected' set amongst the
  %bridge-alterable sets of graphs, and even amongst the bridge-addable sets.
  Consider $R_n \inu \ca$ where $\ca$ is bridge-addable.
  Starting from the lower bound $e^{-1}$ %in~\cite{msw05} 
  on the probability that $R_n$ is connected and the stronger stochastic bound
  $\kappa(R_n) \leq_s 1+ \Po(1)$ on the number of components, 
  it was natural to enquire to what extent the bounds could be improved to match known results for forests.
  Our results suggest that we should think of these bounds as being out asymptotically by a factor~2
  in the `parameter', in that the ratio $\lambda/\nu$ should be doubled
  (though the corresponding bounds are tight in the rooted case).
  
  The central conjecture on connectivity for a bridge-addable set of graphs is from~\cite{msw06},
  and was re-stated here as Conjecture~\ref{conj.b-add}.
  As we noted earlier, some asymptotic improvement has been made on the bound $e^{-1}$
  % on the probabilty that $R_n$ is connected
  \cite{bbg07,bbg10,sn2012}; and the full improvement to $e^{-\frac12}$ has been achieved,
  but only when we make the stronger assumption that $\ca$ is bridge-alterable~\cite{amr2011,kp2011}.

  In the present paper we considered corresponding improvements concerning the distribution of $\kappa(R_n)$,
  and introduced new bounds on $\frag(R_n)$ (the number of vertices left when we remove 
  a largest component)
  which also match results for forests asymptotically.  Further, we set these results in a general framework
  emphasising the role of bridges, rather than just considering uniform distributions.
  
  For random rooted graphs our non-asymptotic results already match those for forests.
  In each other case, to achieve results asymptotically matching those for forests we have had to assume that the set
  $\ca$ of graphs is bridge-alterable.
  It is natural to ask whether these results actually hold under the weaker assumption that $\ca$
  is bridge-addable.  Conjecture~\ref{conj.b-add} is still open. %Let us go back to uniform distributions,
  We propose two further conjectures (for uniform distributions).
  The first concerns a possible extension of inequalities~(\ref{eqn.b-alt-frag2}) and~(\ref{eqn.b-alt-comps1})
  from bridge-alterable to bridge-addable.  
  
\begin{conjecture} \label{conj.b-add-comps}
  If $\ca$ is bridge-addable and $R_n \in_u \ca$ then
 \[ \mbox{(a)} \hspace{.1in} \kappa(R_n) \preceq_s 1+ \Po(\frac12) \;\;\mbox{ asymptotically},\]
  and 
 \[ \mbox{(b)} \hspace{.1in} \limsup_{n \to \infty} \E[\frag(R_n)] \leq 1. \hspace{.9in}\]
\end{conjecture}
  The work of Balister, Bollob{\'a}s and Gerke~\cite{bbg07,bbg10} mentioned earlier
  gives some progress on part (a) of this conjecture: the proofs there
  together with Lemma~\ref{lem.po1} here show that, with $\alpha=0.7983$,
\[ \kappa(R_n) \preceq_s 1+ \Po(\alpha) \;\;\mbox{ asymptotically};\]
  and this bound gives
\[ \limsup_{n \to \infty} \E[\kappa(R_n)] \leq 1+ \alpha \approx 1.7983 \]
  as may be seen by arguing as at the end of the proof of Theorem~\ref{prop.asympt-conn}.
  Note also that to establish part (b) of the conjecture,
  it suffices to show that~(\ref{eqn.frageps}) holds whenever $\ca$ is bridge-addable.
  %(it is less obvious what we learn about $\E[\frag(R_n)]$).
  \smallskip
  
  Finally, we propose a strengthened non-asymptotic version of the last conjecture,
  along the lines of Conjecture 5.1 in~\cite{amr2011} or Conjecture 1.2 of~\cite{bbg10}.
  %We consider $R_n^{\cf} \inu \cf$ where $\cf$ is the class of forests,
  %and compare $\Frag(R_n)$ and $\Frag(R_n^{\cf})$ for each~$n$ (not asymptotically),
  %in terms of numbers of components and of vertices.
  %
\begin{conjecture} \label{conj.b-add-strong}
  If $\ca$ is bridge-alterable, $n$ is a positive integer, $R_n \in_u \ca$ and $R_n^{\cf} \inu \cf$ then
\[ \kappa(R_n) \leq_s \kappa(R_n^{\cf})\;\; \mbox{ and } \;\; \E[\frag(R_n)] \leq \E[\frag(\RF_n)]. \]
\end{conjecture}  
  Establishing the stronger version of this conjecture, in which we assume only that $\ca$ is bridge-addable, would 
  of course be even better!
  
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\bibliographystyle{plain} 






 
\end{document}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
  
%{\small
\begin{thebibliography}{14}

%\m{add ref for rooted forests}

\bibitem{amr2011} L. Addario-Berry, C. McDiarmid and B. Reed.
  \newblock Connectivity for bridge-addable monotone graph classes,
  {\em Combinatorics, Probability and Computing}, first view online, July 2012.
  %2011, arXiv:1110.0009.

 %\bibitem{amr08} \m{update}
 % L. Addario-Berry, C. McDiarmid and B. Reed,
 % Connectivity of Addable Monotone Graph Classes, 2008.
  %submitted 12-05-08 to CPC
    
\bibitem{bbg07}
  P. Balister, B. Bollob{\'a}s and S. Gerke,
  {Connectivity of addable graph classes}. {\em J. Combin. Th. B} {\bf 98} (2008) 577 -- 584.
  
\bibitem{bbg10}
  P. Balister, B. Bollob{\'a}s and S. Gerke,
  Connectivity of random addable graphs, {\em Proc. ICDM 2008} No 13 (2010) 127 -- 134.
    
%\bibitem{dvw96}
%  A.~Denise, M.~Vasconcellos and D.~Welsh,
%  The random planar graph,
%  {\em Congressus Numerantium }{\bf 113} (1996) 61--79.
  
\bibitem{drmota2009}
  M. Drmota, {\em Random Trees}, Springer, 2009. 
 
%\bibitem{fs09}
%  P. Flajolet and R. Sedgewick,
%  {\em Analytic Combinatorics}, CUP, 2009.
     
% \bibitem{gmsw05}
%   S. Gerke, C. McDiarmid, A. Steger and A. Weissl,
%   Random planar graphs with n nodes and a fixed number of edges,
%  {\em  Proceedings of the ACM-SIAM Symposium on Discrete Algorithms (SODA)}
%  2005, 999 -- 1007.

%\bibitem{gmsw07}
%  S. Gerke, C. McDiarmid, A. Steger and A. Weissl,
%  Random planar graphs with given average degree,
%  in {\em Combinatorics, Complexity and Chance,
%  a tribute to Dominic Welsh} (G. Grimmett and C. McDiarmid eds)
%  Oxford University Press, 2007, 83 -- 102.
  
%\bibitem{gn09a}
%  O. Gim\'enez and M. Noy,
%  Asymptotic enumeration and limit laws of planar graphs.
%  %arXiv: math.CO/0501269, 2005  arXiv: math.CO/0501269.
%  {\em J. Amer. Math. Soc.} {\bf 22} (2009) 309--329.

\bibitem{gn09b}
  O. Gim\'enez and M. Noy,
  Counting planar graphs and related families of graphs,
  in {\em Surveys in Combinatorics 2009}, 169 -- 329, Cambridge University Press, Cambridge, 2009.

\bibitem{grimmett06}
  G. Grimmett, {\em The Random-Cluster Model},
  Springer, 2006.

%\bibitem{jlr00}
%  S. Janson, T. {\L}uczak and A. Ruci{\'n}ski,
%  {\em Random Graphs}, Wiley Interscience, 2000.

 
%\bibitem{kmcd11}
%  M. Kang and C. McDiarmid,
%  Random unlabelled graphs containing few disjoint cycles,
%  {\em Random Structures and Algorithms} {\bf 38} (2011) 174 -- 204.
  
\bibitem{kp2011}
  M. Kang and K. Panagiotou,
  On the connectivity of random graphs from addable classes, 2011.
  %\m{arxiv?}

%\bibitem{km08b}
%  Kurauskas, V. and McDiarmid, C.  (2008)
%  Random graphs with few disjoint cycles.
%  {\em Combinatorics, Probability and Computing}, published online 9 June 2011.
  
%\bibitem{km11}
%  Kurauskas, V. and McDiarmid, C.  (2011)
%  Random graphs containing few disjoint excluded minors. 

%\m{Lovasz, problem 4.3}
%\bibitem{lovaszbook}
%  L. Lov\'asz,
%  {\em Combinatorial Problems and Exercises},
%  North-Holland, 2nd edition, 1993.
  
  
%\bibitem{luczak98}
%  T. {\L}uczak,
%  Random trees and random graphs,
%  {\em Random Structures and Algorithms} {\bf 13} (1998) 485 -- 500. \m{wanted?}

%\bibitem{lp92}
%  T. {\L}uczak and B. Pittel.
%  Components of random forests,
%    {\em Combinatorics, Probability and Computing} {\bf 1} (1992) 35 -- 52. \m{wanted?}

\bibitem{cmcd08}
  C.~McDiarmid,
  Random graphs on surfaces,
  {\em J. Combinatorial Theory B} {\bf 98} (2008) 778 -- 797.

\bibitem{cmcd09}
  C.~McDiarmid,
  Random graphs from a minor-closed class,
  {\em Combinatorics, Probability and Computing} {\bf 18} (2009) 583 -- 599.
  
\bibitem{cmcd-rwg2012}
  C.~McDiarmid, 
  Random weighted graphs from a minor-closed class,
  in preparation.
  
%\bibitem{cmcd-unlab2012}
%  C.~McDiarmid, 
%  Minor-closed unlabelled graph classes,
%  in preparation.
  
%\bibitem{cb08}
%   C.~McDiarmid and B. Reed,
%   On the maximum degree of a random planar graph,
%   {\em Combinatorics, Probability and Computing} {\bf 17} (2008) 591-- 601.

\bibitem{msw05}
  C.~McDiarmid, A.~Steger and D.~Welsh,
  Random planar graphs,
  {\em J. Combinatorial Theory B} {\bf 93} (2005) 187 -- 206.

\bibitem{msw06}
  C.~McDiarmid, A.~Steger and D.~Welsh,
  Random graphs from planar and other addable classes,
  {\em Topics in Discrete Mathematics} (M. Klazar, J. Kratochvil,
  M. Loebl, J. Matousek, R. Thomas, P. Valtr, Eds.),
  Algorithms and Combinatorics 26, Springer, 2006, 231 -- 246.

\bibitem{moon67}
  J.W.~Moon,
 {Enumerating labelled trees}, in
 {\em Graph Theory and Theoretical Physics} (F. Harary, Ed.),
 Academic Press, San Diego, 1967, 261 -- 272. 

\bibitem{moon70}
  J.W.~Moon,
 {\em Counting labelled trees},
 {Canadian Mathematical Monographs} {\bf 1}, 1970.

\bibitem{sn2012}
  S.~Norine, private communication, 2012.
  
%\bibitem{ps10}
%  K. Panagiotou and A. Steger,
%  Maximal biconnected subgraphs of random planar graphs,
%  {\em ACM Transactions on Algorithms} {\bf 6} (2010) art. no. 31.
%  \m{2010?}

\bibitem{pitman99}
  J. Pitman,
  Coalescent random forests,
  {\em J. Combinatorial Theory A} {\bf 85} (1999) 165 -- 193.
    
\bibitem{renyi59}
   A. R\'enyi, 
   Some remarks on the theory of trees,
   {\em Publications of the Mathematical Institute of the
   Hungarian Academy of Sciences} {\bf 4} (1959) 73 -- 85.

\bibitem{renyi1970}
  A. R\'enyi, On the enumeration of trees,
  {\em Combinatorial Structures and their Applications},
  R. Guy, H. Hanani, N. Sauer and J. Schonheim (Eds), Gordon and Breach, New York, 1970, 355 -- 360.
 
\end{thebibliography}
%}  
  
\end{document}

